Contract between the

Brookline School Committee

and the

Brookline Educators Union

2011-2014

UNIT A

Printed on Recycled paper by the Brookline High School Copy Center

SCHOOL COMMITTEE

Rebecca Stone -- Chair Alan Morse* - Vice Chair

Helen Charlupski Elizabeth Childs* Amy Kershaw

Susan Wolf Ditkoff Henry Warren* Judy Meyers

Barbara Scotto

ADMINISTRATION

William H. Lupini Peter C. Rowe

Superintendent of Schools Deputy Superintendent

for Administration and Finance

Jennifer Fischer-Mueller

Deputy Superintendent Kirsten Esposito Balboni

for Teaching and Learning

Assistant Superintendent for Student

Service

Angela F. Allen*

Assistant Superintendent for Human Resources

BROOKLINE EDUCATORS UNION

Jessica Wender-Shubow - President Jody Curran - Vice President David Weinstein - Treasurer Lesley Jarvis - Secretary

NEGOTIATION COMMITTEE

Eric Schiff - Chair

Amie Buchman Jody Curran
Jenna Goodearl Cori Green
Jessica Wender-Shubow David Weinstein

Eleanor Demont Robert Miller

^{*} School Committee Negotiation Team

Table of Contents

Article 1:	General	1
1.1	Purpose	1
1.2	Scope of Agreement	1
Article 2:	Recognition	2
Article 3:	1 1 2	
3.1	Salary Schedules in Appendix A and B	
3.2	Longevity	
3.3	Elective Longevity Buyout ("ELBO")	
3.4	Schedule of Extra Compensation in Appendix B	
3.5	Additional Salary in Recognition of Prior Service	
3.6	Placement on Salary Schedule on Initial Employment	
3.7	Schedule of Payments	
3.8	Payroll Deduction Statements	
3.9	Other Payroll Deductions	
3.10	Group Insurance	
3.11	Tax Sheltered Annuity	
3.12	Yearly Record of Attendance	
3.13	New Positions	11
3.14	Bias-Free Appointments	11
3.15	Visitation Day	11
3.16	Promotions	11
3.17	Transfers	11
3.18	Summer School	12
3.19	Packing	12
3.20	Town Facilities	13
3.21	Tuition Reimbursement Fund	13
Article 4:	Conditions of Professional Service	. 13
4.1	Work Year	
4.2	School Calendar	
4.3	Teaching Hours and Teaching Loads	
4.4	Extra-Curricular Duties	
4.5	Professional Staff-Pupil Ratio.	
4.6	Attendance and Participation in Meetings	
4.7	Registers of Attendance	
4.8	Instructional Aides	
4.9	Teacher Indemnity	
4.10	Health and Safety	
4.11	Class Size Report	
1.11	C1000 D120 100 P011	41

Article 5:	Leaves of Absences	21
5.1	Sick Leave	21
5.2	Sick Leave Bank	23
5.3	Medical Documentation Protocols	24
5.4	Requests for Leaves, Extensions or Renewals	26
5.5	Leaves of Absence with Pay	
5.6	Leaves of Absence without Pay	
5.7	Superintendent's Approval of Leaves	
5.8	Child-bearing Leave.	
5.9	Child-rearing Leave	
5.10	Sabbatical Leave Policy	
Article 6:	Evaluation and Supervision	35
6.1	Evaluation and Supervision	35
6.2	Procedure for Evaluation	36
6.3	Duration of Evaluation System	38
6.4	Withholding Salary Increment	39
6.5	Personnel File	39
6.6	Derogatory Material	39
6.7	Academic Freedom	39
6.8	Professional Development and Educational Improvement	39
6.9	Employee Assistance Program	
6.10	Evaluation of Administrators	40
6.11	Discipline with Discretion	40
Article 7:	Just Cause	41
Article 8:	Conditions of Development and Implementation of Educational Programs	41
8.1	Policy Recommendations	
8.2	Policy Review	41
8.3	In-Service Training	41
	Grievance Procedure	
9.1	Definition of Grievance	
9.2	Time Limits	
9.3	Level One	
9.4	Level Two	
9.5	Level Three	
9.6	Level Four	
9.7	Resolving Grievances	
9.8	Arbitrability	
9.9	General Provisions	
9.10	Legal and Equitable Remedies	43

Article 1	0: BEU Rights and Privileges	43
10.1	Use of Buildings and Facilities	
10.2	Right to Address	43
10.3	Information on New Hires	
10.4	Athletic facilities	44
10.5	BEU Days	44
10.6	Agency Service Fee	
10.7	Meetings with Superintendent	
Article 1	1: Reductions In Force ("RIF")	45
Article 1	2: Sexual Harassment Policy	51
Article 1	3: Duration	52
SIDE LE	TTERS OF AGREEMENT	53
1.	New - Miscellaneous	
2.	Definition of Domestic Partner	
3.	Time and Learning Implementation	
4.	Relicensure	
5.	CORI Agreement	
6.	K-8 Teacher (effective 9/1/07)	
APPENI	DIX A: Salary Schedules	57
A.1	Teachers' Salary Schedules	57
A.2	Library Assistants' Salary Schedules	60
A.3	Lab Specialists' Salary Schedules	62
	OIX B: Extra Compensation	
B.1	Compensatory Time	
B.2	Extra Compensation for Specific Positions	
B.3	AP Science Teachers	
B.4	Intramural After School Sports Program	
B.5	Mileage Allowance	
B.6	Additional Funds for Coaching	
B.7	Other Extra Compensation	
B.8	Coaches and Advisors	68
	DIX C: Guidelines	
C.1	Class Size	
C.2	Teacher Facilities	
C.3	Implementation of Educational Programs	73
APPENI	DIX D: IPDP	74
A DDENIE	OIV E. Alternative Evaluation Process: Professional Growth Project	77

CONTRACT

This contract, dated as of September 1, 2011, is between the School Committee of the Town of Brookline ("BSC") and the Brookline Educators Union ("BEU") on behalf of Unit A as described in Section 1.2 and is made by the parties pursuant to Chapter 150E of the General Laws.

This contract includes Appendices A, B, C D and E, which are attached hereto and made part hereof. This contract incorporates the Sexual Harassment Policy and the Evaluation Principals and Format Appendix found under separate cover.

Article 1: General

1.1 Purpose

The purpose of this contract is to establish procedures by which the BSC and the BEU may work together to maintain and improve the quality of the educational program of the Brookline schools, to negotiate teachers' salaries and conditions of employment, to ensure that the professional staff follows procedures and meets standards of performance which are satisfactory to the BSC, and to ensure the right of the professional staff to negotiate and to discuss with the BSC matters which may be of concern to individual teachers or to groups of teachers.

1.2 Scope of Agreement

- A. A conflict of one part or provision of this contract with any law shall not thereby affect the validity or applicability of any other part or provision of this contract.
- B. For purposes of collective bargaining with respect to wages, hours, standards of productivity and performance, and other conditions of employment, the negotiation of agreements and any questions arising there under, the members of the professional staff covered by this contract shall include all professional employees, excepting the Superintendent, the Deputy Superintendents, the Assistant Superintendent, the doctors and substitute nurses, the Headmaster, the Assistant Headmaster, the BHS Deans, the Principals, the Grants Coordinator, the Supervisor of Adult and Community Education, administrators in Bargaining Unit B, substitutes, summer school staff, and instructional aides. As used in this contract, the words "professional staff" and "teachers" shall be synonymous and shall include all professional employees except as the context may otherwise require or in accordance with the coverage of the Agreement described above; with respect to nurses, occupational therapists, and physical therapists, references to "teachers with professional status" shall be deemed to refer to nurses, occupational therapists, and physical therapists who have completed at least three continuous years of service in the Public Schools of Brookline; and references to teachers who have not acquired professional teacher status (herein called "teachers with pre-professional status") shall be deemed to refer to nurses, occupational therapists, and physical therapists who have not completed three continuous years of service in the Public Schools of **Brookline**

- C. The BSC will treat teachers who have worked for more than three (3) previous full consecutive school years in the Public Schools of Brookline as teachers with Professional Teacher Status under the parties' contract to the extent allowed by law. The parties agree not to implement this provision in violation of the law.
- D. The BSC agrees to recommend that the Superintendent of Schools consider granting PTS to teachers who, as of the June 17, 2004, have worked more than four (4) previous full consecutive school years as teachers for the Public Schools of Brookline and who have worked in the previous full school year in a position for which such teachers have certification.
- E. The position of Lab Specialist (e.g. Foreign Language Lab Specialist, Computer Lab Specialist) shall be covered by the same provisions of this contract which cover the library assistants; however, this is not to be construed as any acknowledgement by either party as to the applicability or non-applicability of any specific provisions of this Agreement.
- F. After the execution of this contract by both parties, the BSC and the BEU shall be responsible for reproducing a sufficient number of copies so that the BEU can distribute a copy to each member of the professional staff.

Article 2: Recognition

- 2.1 The BEU recognizes that under the laws of the Commonwealth of Massachusetts, the BSC as the elected representatives of the citizens of Brookline has the final responsibility for establishing the educational policies for the schools and that the Superintendent of Schools has the responsibility for carrying out the policies so established. The BEU further recognizes its professional responsibilities and commitment to education of the highest quality for the children of Brookline.
- 2.2 The BSC recognizes the BEU in accordance with the provisions of Chapter 150E of the Massachusetts General Laws as the exclusive representative of all professional school personnel as defined in Article 1, Section 1.2.

Article 3: Compensation and Teacher Employment

3.1 Salary Schedules in Appendix A and B. The salary schedules for the professional staff are set forth in Appendix A and B.

3.2 Longevity

Any teacher who has reached the maximum step in the Teachers' Salary Schedule and has completed more than 10 full years of continuous service (as defined in Article 11.1) in the Public Schools of Brookline as of September 1 of a given year shall have his/her annual salary, as indicated by the appropriate schedule for that year, increased by the amount listed below:

Years of Continuous	Eff. 9/1/10	Eff. 9/1/12	Eff. 9/1/13
Service Completed	<u>Amount</u>	<u>Amount</u>	<u>Amount</u>
More than 10 but less than or = to 14	\$ 927	\$945	\$964
More than 14 but less than or = to 19	\$ 1,185	\$1,208	\$1,233
More than 19 but less than or = to 24	\$ 1,957	\$1,996	\$2,036
More than 24	\$ 2,215	\$2,259	\$2,305

If any such teacher is employed less than full-time in any such year, the additional salary set forth in the preceding paragraph shall be prorated accordingly.

Unit A personnel not on the Teachers' Salary Schedule are considered at the maximum for the purposes of longevity when they have served enough years that they would be on maximum had they been on the Teachers' Salary Schedule based on their initial placement and years of service.7

3.3 Elective Longevity Buyout ("ELBO")

The entire ELBO and all attendant language was sunset as of August 31, 2008 and can be referenced in Article 3.3 of the 2009-2011 BEU Unit A Contract.

3.4 Schedule of Extra Compensation in Appendix B: The schedule of extra compensation for work performed outside of the workday is set forth in Appendix B.

3.5 Additional Salary in Recognition of Prior Service

- A. Effective August 31, 2010 sunset Article 3.5A in its entirety.
- B. All full-time and part-time employees who were actively employed during the entire 1991-92 contract year shall receive payments when their employment is terminated, subject to the following additional conditions: Payments shall be made upon termination to the extent permitted by applicable laws; payment shall be 3% of the applicable rate in effect when the employee ends his/her employment with the BPS; the applicable rate shall be based upon the column and step (excluding longevity and all other premiums) which the employee occupied during the 1991-92 contract year; if the employee was less than full time in the 1991-92 school year, such payment shall be pro-rated accordingly; payment shall be made only to those teachers who have completed at least 10 full years of service in the BPS as of the date of their termination; and the employee shall give written notice (or via e-mail) to the Superintendent as soon as possible and no later than April 1 of his/her intent to terminate his/her employment at the end of that year. Such notice will be waived by the Superintendent when failure to give such notice is beyond the reasonable control of the teacher.

Laid off employees with recall rights shall be eligible when their recall rights expire (if they have not been recalled). In such cases, the applicable rate shall be the rate in effect in the last contract year in which the employee worked. In the event of an employee's death any monies due under this section shall be paid to his/her estate.

3.6 Placement on Salary Schedule on Initial Employment and Thereafter

- A. Employees hired in this unit will be initially placed on the salary scales as follows. One-half or greater fraction of a year of experience shall be counted as one year.
 - i. A teacher, on initial employment, shall be placed on the salary schedule according to previous teaching experience and degrees and credits earned at accredited institutions.
 - ii. A nurse, library assistant or lab specialist, on initial employment, shall be placed on the salary schedule according to previous experience in this field and degrees and credits earned at accredited institutions.
 - iii. Librarians and non-classroom teaching specialists not listed in paragraph ii above, on initial employment, shall be placed on the salary schedule according to degrees earned at accredited institutions and previous experience in a school setting, or after being licensed or certified or earning a masters degree, previous experience working with school-aged children. Librarians will also receive credit for previous experience in public libraries.
 - iv. Effective September 2012, the term "experience" in subsections i, ii, and iii above shall not include:
 - a. Experience outside of Massachusetts;
 - Experience in a position in or outside of Massachusetts for which a license issued by the Massachusetts Department of Elementary and Secondary Education (DESE) was not required by DESE;
 - c. Experience when the employee did not hold a Massachusetts DESE license.
 - v. Notwithstanding any provision to the contrary in this Agreement, the Superintendent or his/her designee shall have the sole discretion to place an individual upon initial employment or reemployment in the Brookline Public Schools on the salary schedule at any step in the appropriate lane. The Superintendent/designee shall notify each such individual of his/her placement under this provision. Such discretion shall be limited to no more than ten (10) individuals per academic year.

B. Column Movement

i. A teacher whose qualifications have been determined to qualify him/her to move from one level of preparation to the next higher level (e.g. B-M, M-M+30, etc.) shall be placed on the higher salary scale at the step he/she would have been on as if he/she was initially placed on that preparation level, upon presentation to the Superintendent of satisfactory evidence that the teacher is entitled to the change and upon approval by the BSC. Column movement will be implemented following receipt by the Superintendent of such documentation and shall be

- retroactive to the date that the institution states the work was completed, or retroactive up to 30 days from receipt of such documentation by the Superintendent, whichever is later. Such change may be made at any time if the teacher has notified the Superintendent in writing (or via e-mail) by January 15th of the preceding school year of the expected change and date thereof; otherwise, such change shall be made at the commencement of the next school year.
- ii. The Status of Master's Degree + 30 is awarded to any teacher who has completed 30 semester hours of graduate work approved in advance by the Superintendent of Schools. No equivalents are permitted. The Status of Master's Degree + 45 is awarded to any teacher who has completed 45 semester hours of graduate work approved in advance by the Superintendent of Schools. No equivalents are permitted.
- iii. The employee must obtain a final grade of a "B" or better in a graduate-level course(s) from an accredited college and/or university, which would count for graduate degree credit at such institution and which has been pre-approved by the Superintendent or his/her designee as acceptable toward the semester hours for movement from the Master's degree lane to the Master's Degree + 30 lane and/or the Master's Degree + 45 lane or from the Master's Degree + 30 lane to the Master's Degree + 45 lane. Graduate-level courses taken on-line from an accredited college and/or university which would count for graduate degree credit at such institution and which have been pre-approved by the Superintendent or his/her designee as acceptable toward the semester hours for lane movement shall be treated the same as other graduate-level courses provided the employee has earned a grade of B or better in the course. An employee must request and obtain prior approval from the Superintendent or his/her designee to take a course on a pass/fail basis if the employee wishes to have the semester hours from such a course count toward the semester hours for lane advancement. In considering requests for advance approval of graduate work, the Superintendent or his/her designee may consider the relevancy of the graduate work to the employee's assignment, license attainment, and/or license renewal.
- iv. In the absence of earning graduate credit, National Board Certification counts for six (6) credits.
- v. Effective September 1, 2006, a JD or LLM shall count as a Masters degree. Subsequent graduate work, including earning a Masters degree, shall count toward column movement. This provision does not apply to employees who as of August 31, 2006 are being paid on the Doctorate Column.
- vi. Effective September 2012, each Unit A member will be provided with a written or email statement prior to the first day of school with the following information: assigned grade level or subject for the start of the school year, assigned school(s), salary schedule placement (step and lane), salary, and extra compensation where

vii. applicable and known.

- C. Effective September 1, 2012, the Bachelor's lane shall be restructured as follows:
 - i. All steps beyond Step 5 shall be eliminated for employees who commenced their employment in Unit A on or after September 1, 2012 with the Public Schools of Brookline.
 - ii. Employees who commenced employment in Unit A in or before the 2011-2012 school year and who were on Step 5 or below in the 2011-2012 school year shall not progress past Step 5 in the Bachelor's column.
 - iii. Employees who commenced employment in Unit A in or before the 2011-2012 school year and who were on Step 6 or above during the 2011-2012 school year may continue through the Steps in effect in the Bachelor's column.
 - iv. Nurses, automotive teachers, construction teachers and food services teachers who are not required to have a degree above the Bachelor's level may progress through all steps in the Bachelor's column.
 - v. Effective September 1, 2013 add a new Bachelor's Step 14 at the rate of \$2,500 above Step 13.
 - vi. Effective at 11:59 p.m. on August 31, 2014 add a new Bachelor's Step 15 at the rate of \$2,500 above Step 14.
- D. Effective 12:05 a.m. on September 1, 2013, add a new top step at \$2,500 above the prior top step in each of the following columns: Master's Degree, Master's Degree +30, Master's Degree +45, and Doctorate Degree.

Effective at 11:59 p.m. on August 31, 2014, add a new top step at \$2,500 above the prior top step in each of the following columns: Master's Degree, Master's Degree +30, Master's Degree +45, and Doctorate Degree.

Effective September 1, 2013, eliminate Step 1 from each of the following columns: Master's Degree, Master's Degree +30, Master's Degree +45, and Doctorate Degree.

Double Step: Effective September 1, 2013, employees who have a Master's Degree or above who have completed seven (7) consecutive work years of service with the required license in Unit A in the Public Schools of Brookline <u>and</u> who are beginning the eighth consecutive work year of licensed service shall take a double step. (The Double Step shall not apply to any employee who is beyond his/her eighth consecutive year of service as of September 1, 2013.) Unpaid leaves of absence pursuant to Section 5.6 of Leaves of Absence without Pay represent a break of continuous service for the purposes of this Double Step. A leave of absence of more than one year pursuant to Section 5.9 Child Rearing Leave shall constitute a break in service for the purposes of this Double Step. A leave of absence of up to <u>one</u> year

pursuant to Section 5.9 Child Rearing Leave shall not constitute a break in service for the purposes of this Double Step but said leave shall not count towards satisfying the seven consecutive work year requirement.

Examples:

- 1. Teacher A has a Master's Degree. She has served as a teacher in the Public Schools of Brookline for seven (7) consecutive years with the requisite license for her position. She was at Step 7 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is commencing her eighth consecutive year of service with the requisite license. She takes a double step and is placed on Step 9.
- 2. Teacher B has a Master's Degree. She has served as a teacher in the Public Schools of Brookline for 10 years with the requisite license for her position. She was at Step 10 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is commencing her 11th consecutive year of service. She is placed on Step 11.
- 3. Teacher C has a Master's Degree. She served as a teacher in the Public Schools of Brookline for six (6) years then took a leave of absence (other than for Child Rearing pursuant to Section 5.9) for one year without pay. She then returned to work and has worked for one year (2012-2013 academic year). At all times she had the requisite license for her position. She was at Step 7 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is on Step 8.
- 4. Teacher D has a Master's Degree. She has served as a teacher in the Public Schools of Brookline for seven (7) consecutive years. She worked under a waiver for her first year in the District and has had the requisite license for her position starting with her second year. She was at Step 7 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is commencing her eighth consecutive year of service. She is placed on Step 8. On September 1, 2014, she is commencing her ninth consecutive year of service and placed on Step 10 because she has been licensed for seven (7) consecutive years.
- 5. Teacher E has a Master's Degree. She served as a teacher in the Public Schools of Brookline for six (6) years then took a leave of absence for one year without pay for Child Rearing pursuant to Section 5.9. She then returned to work and has worked for one year (2012-2013 academic year). At all times she had the requisite license for her position. She was at Step 7 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is on Step 9.

- 6. Teacher F has a Master's Degree. She served as a teacher in the Public Schools of Brookline for six (6) years then took a leave of absence for two (2) years without pay for Child Rearing pursuant to Section 5.9. She then returned to work and has worked for one year (2012-2013 academic year). At all times she had the requisite license for her position. She was at Step 7 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is on Step 8.
- E. Teachers employed half-time or greater will be advanced on the salary schedule in the same way as full-time staff. Teachers employed less than half-time will serve two years before advancing one step on the salary schedule.
- F. In the initial employment of a teacher, service in the Peace Corps, the Commonwealth or National Teacher Corps shall be counted for teaching experience when such service was related to the teaching duties to be assumed.
- G. The salary of employees who are employed less than full-time shall be prorated in accordance with existing practices.
- 3.7 Schedule of Payments. Salary payments to all teachers shall be made biweekly and delivered in envelopes, the first payment to be made on the second Thursday after Labor Day. Payments will be made in twenty-one (21) installments. Each teacher shall receive on or before May 1 of each school year a checklist which sets forth all obligations he/she is expected to complete by the close of the school year.
- 3.8 Payroll Deduction Statements. Payroll deduction statements showing earnings, biweekly deductions, and usual year-to-date deductions shall be issued with each paycheck. The direct deposit program implemented by the Town of Brookline shall apply to employees covered by this Agreement. Effective with the first pay period in the 2010-2011 work year, all employees represented by the BEU shall receive their pay through direct deposit; the BSC or its designee will provide direct deposit notification statements either electronically or in paper form to employees. (The parties understand and agree that employees will need to provide direct deposit information to the payroll office sufficiently prior to the start of the 2010-2011 work year to enable the BSC to set up direct deposit for their paycheck and directly deposit their first pay check of the 2010-2011 work year.)
- **3.9 Other Payroll Deductions**. The BSC shall provide that, whenever duly authorized by any teacher, payroll deductions on behalf of such teacher shall be made every payday and paid over in accordance with such authorization for any or all of the following:
 - 1) Purchase of U.S. Savings Bonds
 - 2) Local, State, and National Dues or agency service fee
 - 3) Payments to Brookline Credit Union
 - 4) Premiums under annuity contract
 - 5) Premiums under Employees Group Insurance Plan
 - 6) MTA Credit Union
 - 7) BEU Supplemental

As it relates to the BEU Supplemental, the BEU will furnish a list of employees who have authorized a deduction together with the specific amount of the deduction to the payroll office. Changes to that list may be done quarterly. The payroll office will then issue to the BEU a check each pay period for the conglomerate amount of all deductions in this category for all members who have authorized such deduction, along with a list of members for whom the deduction was made.

The parties agree that the check off for union dues shall apply to part-time teachers when and to the extent the new payroll service permits.

Upon the teacher's voluntary written assignment filed at least three weeks in advance, the BSC agrees to deduct from the teacher's salary amounts due and payable to the Brookline Credit Union and/or the MTA Credit Union as authorized by the teacher. The amount so deducted shall be forwarded to the appropriate credit union in accordance with existing procedures. Any such deductions shall be terminated upon the written request of the teacher filed at least three weeks in advance of the effective date of termination, in which case the BSC will not be required to honor another voluntary assignment filed by the teacher until the following school year. If the Town Treasurer notifies the BSC that administration of this section will cause a significant administrative problem for the Town, the BSC will so notify the BEU, and it is agreed that the provision for the MTA Credit Union deduction shall be terminated at the end of that school year.

The Town currently permits employees to authorize payroll deductions for the purpose of making "pre-tax" contributions pursuant to Section 125 of the IRC ("Section 125") for certain benefits, e.g. employee contributions to group health insurance plans. If the Town enters into an agreement with a Section 125 administrator and if the agreement permits the employees to purchase additional services/products through such administrator which are not covered by Section 125, the Town will make available the opportunity for payroll deductions for employees to purchase such services/products through the administrator.

3.10 Group Insurance

- A. Health Insurance is addressed through the Public Employee Committee ("PEC") agreement between the Town and the PEC. This Section A is included for informational and historical purposes only.
- B. This section has been intentionally left blank.
- C. The BSC shall continue the insurance coverage of any teacher who is on leave with pay (including sabbatical leave). Teachers who are on leave without pay shall be entitled to continue their participation in the group insurance plan, provided they pay 100% of the premium associated with such participation in accordance with a schedule established by the BSC.

- D. The BSC shall make its payments of insurance premiums on the policies set forth above to provide insurance coverage for the full 12-month period commencing September 1 and ending August 31; when necessary, premiums on behalf of the teacher shall be made retroactively or prospectively to assure uninterrupted participation and coverage.
- E. Life insurance coverage shall be \$5,000. It is understood that the Town shall pay 75% of the premium cost of such coverage and that each participating employee shall pay the remaining 25%.
- F. Health Insurance Coverage for Domestic Partners. If Massachusetts law governing health insurance which may be extended by municipalities to their employees (e.g., M.G.L. ch. 32B) is amended to permit coverage of domestic partners other than spouses, or if Massachusetts law is changed to permit such coverage if adopted by an appropriate municipal authority, the BSC will provide such coverage as soon as is practicable or, as the case may be, will request that the appropriate municipal authority adopt the legislation.
 - It is understood that the BEU neither acknowledges nor admits that Ch.32B as it is presently constituted in any way disallows such coverage.
- G. The premium conversion plan under Section 125 of the Internal Revenue Code with respect to the health insurance contributions made by employees; the dependent care plan (DECAP) for those employees who wish to participate; and the plan covering un-reimbursed medical expenses (MEDCAP) for those employees who wish to participate shall be made available and administered by the Town of Brookline on the same bases as are applicable to all other employees of the Town.
 - The maximum account for MEDCAP is currently \$1000 per fiscal year. If the Town suffers no net loss in the balances of its MEDCAP accounts for FY 1995, the maximum for FY 97 shall be increased to \$3000. Unless otherwise agreed the same procedure shall be followed for subsequent years i.e., if the Town suffers no net loss in the balances of its MEDCAP accounts in FY 96 the maximum for FY 98 shall be \$3000: If the Town does suffer such a loss in FY 96, the maximum for such accounts in FY 98 shall be \$1000.
- **3.11 Tax Sheltered Annuity**. Teachers will be eligible to participate in the tax-sheltered annuity plan, pursuant to appropriate Federal, state, and local law.
- **3.12 Yearly Record of Attendance**. A yearly Record of Attendance shall be furnished each teacher by September 30 for the preceding school year. This record will include absence for:

Sick Leave (personal)
Administrative Leave (Section 5.4.A)
Death in Family
Funeral
Court Appearance

Military reserve Summer School Other reasons (without salary)

- 3.13 New Positions. During the period while this contract is in effect, the BSC may establish new professional positions and determine the salaries pertaining thereto. The Superintendent shall promptly notify the BEU upon the creation of any new position under this paragraph. If the BSC changes a position, as distinguished from creating a new position, the BSC will negotiate the salary for such changed position with the BEU and will provide a job description to the BEU.
- **3.14 Bias-Free Appointments**. Any appointment, promotion or transfer shall be made without regard to race, creed, color, religion, nationality, sex, sexual orientation, age, marital status and, in accordance with applicable laws, handicap/disability.
- **3.15 Visitation Day.** A candidate shall, if possible, spend at least one visitation day in his/her prospective school or department before being hired.
- 3.16 Promotions. All qualified professional employees will be given adequate opportunity to apply for any positions normally considered to be above their rank and status. An adequate opportunity to apply for such promotion means that during the school year the entire professional staff shall be promptly notified in writing (or via e-mail) that a vacancy exists and that any professional employee applying for promotion shall be given full opportunity to interview by persons designated by the Superintendent. The notice shall contain a job description of the vacancy. If the job description or level of compensation for a posted position is changed prior to being filled, the Superintendent or his/her designee will repost the position noting the change(s).

During the summer vacation, the Assistant Superintendent for Human Resources will give written notice (or via e-mail) of such a vacancy to the BEU President and to any person who has notified the Assistant Superintendent for Human Resources in writing (or via e-mail) that he/she wishes to be informed of such a vacancy. The Assistant Superintendent for Human Resources will also post a notice of such vacancy in the BSC offices. The postings made pursuant to this section shall be for at least 10 days. Pending the permanent filling of the position, the Superintendent or his/her designee, may assign anyone to the position temporarily. In the matter of filling such vacancies, the decision of the Superintendent or his/her designee, shall be final.

3.17 Transfers

A. **Voluntary Transfers**. Any person desiring a transfer shall submit a written request (or via e-mail) to the Superintendent and the building Principal not later than March 1. Requests shall specify the location or the assignment desired. If a suitable opening exists, any necessary interviews will be arranged. The Superintendent shall endeavor to process requests for transfers as promptly as possible and the proposed transfers must be agreeable to the building Principals involved.

Any person who informs the Assistant Superintendent for Human Resources in writing (or via e-mail) that he/she wishes to be notified of a specific vacancy in a position not normally considered to be above his/her rank and status shall be notified if and when such a vacancy occurs. The Assistant Superintendent for Human

Resources will honor any request for confidentiality which is contained in such a notice. During the summer vacation period, the Assistant Superintendent for Human Resources will notify the BEU President, in writing (or via e-mail), and will post a notice at BSC offices concerning any such vacancy.

- B. **Involuntary Transfers**. Written notice of a proposed involuntary transfer and the reasons therefore shall be given to the teacher involved. The transfer shall not be made until the teacher or the teacher and a representative of the BEU have had an opportunity to discuss the proposed transfer with the Superintendent or his/her designee, nor shall such transfer be effected without the consultation of the building principals involved. Upon request of the teacher, a BEU representative may be present at meetings with the teacher concerning involuntary transfers. A teacher being involuntarily transferred may at his/her option take a leave of absence without pay for one (1) year.
- C. **BSC's decision final**. The BSC's decision with respect to any transfer, whether voluntary or involuntary, shall be final; provided that involuntary transfers shall not be used as a form of discipline.
- **3.18 Summer School.** All openings for summer school positions shall be adequately publicized by the Superintendent in each school building. Acknowledgment of applications shall be given within a reasonable time.
- **3.19 Packing.** If a member of this bargaining unit is asked to pack school materials and supplies, the following criteria will be used:
 - A. Each supervisor will jointly determine with each employee in his/her department an agreed upon number of hours for packing. In case of disagreement, the supervisor will make the determination. Once an agreement is established, the employee is free to perform the task at his/her reasonable schedule. These hours are for both packing and unpacking from a staff member's move.
 - B. Reimbursement is only to pack boxes and throw old materials away. Employees will not be reimbursed to review items in detail. Employees are not expected to move boxes or furniture and should use reasonable caution in the packing process to avoid injury. Any individual with a pre-existing concern over potential injury should opt out of the packing process rather than risk injury.
 - C. Employees will be compensated for this work at the workshop rate (see Appendix B.7). Payment will be made within 60 days of an Administrator's submittal of hours to his/her supervisor or designee.

- D. Work compensated at the workshop rate is expected to take place outside of normally assigned duties and release time.
- E. The Superintendent or his/her designee may choose to relieve employees of any duty (e.g. teaching, administrative) in order to do this work instead of paying the workshop rate.
- F. Any employee may choose to not pack or unpack in which case the supervisor will arrange for the packing and moving of materials.
- **3.20 Town Facilities**. Members of this bargaining unit may use Town Facilities on the same terms and conditions as other employees of the Town of Brookline.
- **3.21 Tuition Reimbursement Fund.** Effective September 1, 2007, the BSC will establish a tuition reimbursement fund for employees in positions represented by Unit A in the Public Schools of Brookline. Such fund shall be \$10,000 per contract year. Effective September 1, 2010, such fund shall be \$19,000 per contract year. Unused funds shall not carry over to the next contract year. The following rules shall apply to the Tuition Reimbursement Fund:
 - A. An applicant applying for tuition reimbursement must have been employed by the Public Schools of Brookline in a position represented by Unit A for a minimum of one year prior to applying for tuition reimbursement.
 - B. There is a limit of \$500 of reimbursement per contract year per employee up to the \$10,000 fund maximum per contract year. Effective September 1, 2010, there is a limit of \$600 of reimbursement per contract year per employee up to the \$19,000 fund maximum per contract year.
 - C. The School Department will establish three deadlines for applications for reimbursement (one deadline for each of the following three semesters: fall, winter/spring, and summer).
 - D. Applicants must obtain prior written approval for the course for reimbursement from the Superintendent or his/her designee prior to taking the course; the decision of the Superintendent/designee is final and binding.
 - E. Applicants may not obtain reimbursement for tuition, if there is a voucher available to use for the course.
 - F. The Applicant must obtain a grade of "B" or better in the pre-approved course to be entitled to reimbursement. (If the course is not offered for a grade and is only offered on a pass/fail basis, the employee must receive a pass in the course to receive reimbursement.)

Article 4: Conditions of Professional Service

4.1 Work Year

- A. The scheduled school year shall consist of 188 days. The actual number of days that schools are open for children shall not exceed 180. Teachers shall be present for 183 days, two of which shall occur before the beginning of classes in the Fall, and one for the purpose of professional development.
- B. The professional development day will not be scheduled before the first day or after the last day for students. The activities for this day will be planned by the Deputy Superintendent for Teaching and Learning in collaboration with the Professional Development Council. Effective September 2013, the Professional Development day may be converted to a minimum of six (6) hours of pre-approved professional development learning activities related to a goal(s) of the Public Schools of Brookline. The administration will publish a list of specific learning opportunities that may be used to fulfill this requirement and employees may select from this list or propose to participate in and seek approval for other learning opportunities not included on such list. All professional development activities used to satisfy the 6-hour minimum requirement must be pre-approved by the employee's appropriate supervisor(s). The Superintendent has the authority to suspend the conversion of the professional development day into hours for any year or years and require employees to attend a district professional development day in such year(s).

The conversion of the professional development day into hours shall not affect the per diem rate for employees. The per diem rate shall be calculated by dividing the annual salary in Appendix A by 183 days.

- C. The work year for Lab Specialists shall be 191 days.
- D. Newly employed teachers shall also be present for two additional orientation days prior to the Labor Day weekend. Teachers shall be permitted to work in their classrooms or buildings prior to the opening of school for as many hours or days as the building is open, and to leave at the end of the school year as soon as the individual teacher and his/her Principal or the Headmaster determine that all professional responsibilities have been fulfilled.
- E. To the extent that a guidance counselor's assignment or duties may require, a guidance counselor may work beyond the number of days specified for teachers in Article 4.1, up to a maximum of eight (8) additional days as approved by the Assistant Superintendent for Student Services. Notwithstanding the provisions of Appendix B, Section B.1, not more than half the additional days worked may be taken as compensatory time.
- 4.2 School Calendar. Unless otherwise mutually agreed, the work year for teachers shall not begin before September 1 and shall not end after June 30. The Superintendent shall give notice to the BEU of the proposed calendar no later than one week before the December vacation, and, upon request, the BEU and the Superintendent shall negotiate in good faith before the calendar is publicly distributed or adopted; provided that in the event no agreement is reached by the last regularly scheduled BSC meeting prior to the February vacation the BSC then may adopt and publish the calendar. It is understood that the school calendar shall comply with applicable laws and regulations.

4.3 Teaching Hours and Teaching Loads

A. Teaching Hours

- i. The work day for full-time teachers at the high school will be 6.833 continuous hours between 7:25 am and 3:00 pm. Instruction time will begin no earlier than 7:45 am. Teachers will be present in the building 20 minutes before the start of their first scheduled period. On days when a teacher on an early schedule has an after-school faculty, departmental or IEP meeting, he or she may begin such work day at 7:45 am.
- ii. The work day for full-time teachers in grades K through 8 will be 6.333 continuous hours between 7:30 am and 3:00 pm
- iii. The work day for High School and elementary school librarians will be 7.583 continuous hours between 7:45 am and 4:00 pm. Full-time Library Assistants will be on duty for 8 continuous hours between the hours of 7:45 am and 4:00 pm, including a 30-minute duty-free lunch.
- iv. School adjustment counselors/social workers shall maintain office hours from 8:00 am to 4:00 pm.
- v. The work day for nurses will be 6.833 continuous hours between 7:25 am and 3:00 pm, with a 30-minute lunch break during which nurses are on call for emergencies.
- vi. The work day for the Pre-Kindergarten teachers will be 5.383 continuous hours between the hours of 7:30 am and 2:00 pm or between 11:00 am and 5:30 pm. Four continuous hours of that time will be with students.
- vii. The work day for the Pre-School teachers will be 4.216 continuous hours between the hours of 7:30 am and 1:00 pm or between 11:30 am and 5:00 pm. Three continuous hours of that time will be with students.
- viii. The work day for full-time Pre-School/Pre-Kindergarten teachers will be the same as full-time grade K-8 teachers.
- ix. The work day for Lab Specialists will be 8 continuous hours between the hours of 7:30 am and 4:00 pm, including a 30-minute duty-free lunch.
- x. The work day for occupational therapists and physical therapists will be 6.833 continuous hours between 7:25 a.m. and 4:00 p.m. with a 30-minute duty free lunch.
- xi. Effective September 2013, notwithstanding any provision of this Agreement to the contrary, the instructional day for students and the work day for teachers shall be increased by ten minutes at the end of the teacher's work day on four days per

week and students shall be dismissed forty (40) minutes early on one day per week. The Superintendent or his/her designee shall notify the BEU president on or before March 31st of the prior school year which days shall be extended by 10 minutes and which day shall be an early release day for students. The time after students are dismissed on the early dismissal day shall be used for teacher collaboration in grade-level or grade-span meetings, subject-specific meetings, meetings regarding specific students, meetings designed to further the established goals of the individual teacher, department, school and/or system and (on a more limited basis) other collaboration related activities (i.e. contacting collaborators or summarizing the results of a grade level or subject area meeting.) Such collaboration shall be with other teachers including specialists, ELL teachers, special educators and others and may include participation by Unit B administrators and/or other administrators. Teachers may be asked to report such collaborations to their building principal.

- B. Assignment of individual employees to work day starting time. The assignment of individual employees to work day start times for the following school year will be in accordance with the following timelines and provisions.
 - i. March. The headmaster or designee will circulate a form (in elementary schools the teacher will send a letter to the principal) on which employees will indicate their preference, if any, for start time, including whether there are compelling personal circumstances for not being assigned to a particular start time. Employees may communicate oral information on the nature of the compelling personal circumstance.
 - ii. **May June**. Based on information then available, the headmaster or designee will give to employees notice of the assignment of their start times by June 1 (for elementary principals, 10 days prior to the last school day).

As soon as practical but not more than 10 days after receipt of notice of start time, an employee may submit a request for reconsideration based on a compelling personal circumstance previously communicated by such employee to the headmaster/principal or designee. Information serving as the basis for such request will remain private with the BSC and the BEU.

The BEU and the headmaster/principal or designee will jointly review and try to resolve the requests for reconsideration prior to final assignment.

Final assignment of start dates will be issued by the headmaster/principal by the end of the teacher work year, subject to subsection iii.

The deadlines in this subsection will be strictly enforced.

- iii. **Summer reassignments**. Late budget decisions, budget rescissions or budget changes made after the end of the teacher work year or employee turnover thereafter, or other circumstances beyond the control of the employer may lead to reassignments of employee start times. Request for accommodation based on compelling personal circumstances which were not considered prior to the end of the teacher work year will be reviewed in good faith by the headmaster or designee in conjunction with the BEU where appropriate.
- iv. **Final disposition**. The parties recognize that the final assignment of start times is a BSC right the exercise of which will not be subject to the grievance and arbitration procedure. Notwithstanding the prior sentence, the BEU and not any individual employee may file an institutional grievance to be submitted to expedited arbitration under the Expedited Arbitration Rules of the American Arbitration Association, claiming that a headmaster/principal acted in bad faith under this section B.

C. Impact on Unassigned Periods

The provisions of Sections 4.3 A., B., and D. neither enlarge nor diminish the rights of the parties under Sections 4.3 H., I., J., and K. It is understood and agreed that each high school teacher shall be scheduled for the equivalent of at least 4 standard periods per week as unassigned periods plus X block, exclusive of the one-half hour duty free lunch period per day. It is further understood and agreed that the schedule

for the 97/98 school year includes some class meetings that are 1½ periods and some that are 4/5 of a period but that each high school teacher shall be assigned a maximum of the equivalent of 20 standard instructional periods and 4 standard periods of other assigned duties.

D. Notification of change in school schedules

In the case of any contemplated systemic restructuring of the schedule at any school, the BEU will be given advance notice and an opportunity to discuss such change, prior to notice of change to individual employees. The headmaster or principal will provide the BEU 60 days advance notice when practicable, but in no event less than 30 days prior to such notice to individual employees.

- E. By mutual agreement of the supervisor and member of the bargaining unit, a schedule may be arranged which differs from those set forth in sections A, B and C above provided that the changed schedule may not involve work which begins prior to 7:00 am or ends after 6:00 pm. It is understood that any such arrangements shall be reviewed and evaluated by the member, his/her supervisor and any other appropriate staff after one year.
- F. Physical education teachers in the elementary schools shall be required to participate in the after-school sports program, except when excused by the BSC or where an individual elementary school physical education teacher had prior contract commitments for other work which precluded his/her taking part in the after-school

sports program. Such teachers as are assigned by the school administration to such program shall be required to work not more than 4 such after school sessions per week. If such elementary school physical education teacher were ill on a day on which he/she would otherwise perform such after-school duties he/she shall receive pay for such day. (See Appendix B)

Any physical education teacher hired after July 1, 1977, (and assigned to the High School) may, at the discretion of the BSC, be required to coach up to two seasons per year.

- G. All teachers will have a 30-minute duty-free lunch period to be taken at the following times.
 - i. In the elementary schools, during the regular student lunch period of their grade level where practicable; and
 - ii. In the High School, in accordance with present practices.
- H. Grades 9-12 teachers shall have a maximum of twenty (20) assigned instructional periods and four (4) periods of other assigned duties (one of which may be a homeroom), except as the BSC has previously voted otherwise.

The BSC may change the policy with respect to those teachers who currently receive .25 credits for each class taught notwithstanding the above provision of this section. Any such changes by the BSC will be in accordance with the letter from the BSC to the BEU dated February 27, 1992.

- I. Each elementary teacher (K-6) will be scheduled for a daily preparation period. In a five-day week, at least four (4) of these preparation periods will be scheduled for at least forty (40) minutes in length, while the fifth will be scheduled for at least thirty (30) minutes in length. Such preparation periods are exclusive of the one-half (1/2) hour duty-free lunch period per day. The employee's supervisor shall have the responsibility to schedule preparation and lunch periods.
- J. Teachers responsible for grades 7 & 8 only, will have a minimum of five unassigned periods, exclusive of the one-half hour duty free lunch period per day and will usually be responsible for a homeroom.
- K. Teachers of art, home economics, industrial arts, music, physical education and specialists (except nurses), and teachers responsible for instruction in more than one school in grades K-8 will have a minimum of five unassigned periods per week exclusive of the one-half (1/2) hour duty-free lunch period per day. The employee's supervisor shall have the ultimate responsibility for scheduling. If employed full-time and assigned to one school, then the employee may be responsible for a homeroom. Travel time, appropriate to the distance, will be given to teachers responsible for instruction in more than one school.

L. Meetings, Conferences and Office Time

- i. Teachers shall be available for student conferences after school at reasonable times to be arranged with the elementary building principal or curriculum coordinator in the High School. A teacher may be in his/her classroom or building for as much time either before or after the hours indicated in the preceding paragraphs as he/she, in his/her professional judgment, determines is necessary to provide adequately for his/her programs and children.
- ii. Notwithstanding the provisions of the preceding sentence, a teacher will also be expected to stay in his/her classroom or building for such additional time as his or her Principal, Curriculum Coordinator or their Supervisor reasonably deems necessary.
- iii. As long as the current grading system remains in effect, there will be four scheduled release day afternoons for parent-teacher conferences twice a year; once in the fall and once in the spring for regular classroom teachers in grades K-6. In addition to the above, each school shall offer a 3-hour evening parent-teacher conference block for grades K-6 in both the fall and spring conference periods.
- iv. **Office time**. In addition to Subsections i, ii and iii, effective September 1, 2002, all full-time Unit A personnel other than library assistants and lab specialists will schedule one 30-minute block of time as school office time one day per week for meetings with students or parents. Such time will be scheduled as follows:
 - immediately following the bargaining unit member's assigned work day; or
 - immediately before the bargaining unit member's assigned work day.

At the high school, office time may be scheduled immediately prior to the teacher's first period, with the teacher's arrival time 20 minutes prior to office time; or office time may be scheduled immediately on arrival, preceding 20 minutes before the teacher's first period. Elsewhere in the system, morning office time shall be scheduled to end immediately prior to the beginning of an employee's work day. Such office time shall be scheduled and published at the beginning of the school year and quarterly thereafter to reflect any schedule change.

M. **Kindergarten**. If on November 1 of a given year a kindergarten class exceeds 22 students, 30 hours per week of aide time will be assigned.

N. Student Load

i. It is the goal of the BSC that no full-time high school teachers in Social Studies, Math or Foreign Language shall have an aggregate student load of more than 115 students. If on November 1, such a teacher has a student load in excess of 115, the Headmaster shall make efforts to reduce such load to 115.

ii. Specialists, including nurses, have the right to union representation in any conversation regarding workload that the specialist or nurse may have with his/her supervisor or other administrator.

O. Special Education Workload

A special education teacher (effective September 1, 2007, add: Adaptive P.E. Teacher, physical therapist, speech therapist, and occupational therapist) may apply to his/her principal or, in the case of the high school, the Special Education Coordinator, for up to six (6) hours per quarter of duty-free time for the purpose of completing required paperwork.

4.4 Extra-Curricular Duties

- A. The BSC and the BEU acknowledge that a teacher's primary responsibility is to teach and that his/her energies should be utilized to this end. The BSC and BEU also recognize that there are additional duties and responsibilities of a teacher which must be fulfilled. The BSC agrees that insofar as is practical, such duties and responsibilities shall be equitably assigned.
- B. The BEU recognizes the right of the Superintendent or his/her designees to assign teachers to attend extra-curricular activities, such assignments to be made in an equitable manner.
- C. Part-time teachers shall not be required to perform non-teaching duties to an extent greater than that fraction of a full-time teaching load, rounded off to the nearest full period, for which they have been hired.
- **4.5 Professional Staff-Pupil Ratio**. During the term of this Agreement, the BSC will make all reasonable efforts to maintain the 1972-73 ratio of number of members of this bargaining unit and Unit B to the number of students. Prior to submission of the budget to the BSC, the BEU is invited to consult with the administration to discuss the BEU's priorities in the changing of staffing patterns.

4.6 Attendance and Participation in Meetings

A. Attendance and participation in meetings called by appropriate school officials are professional responsibilities. Reasonable effort shall be made to avoid scheduling more than 38 meetings for any teacher. An agenda shall be sent to each participant three school days before the meeting. Although under normal circumstances meetings shall not exceed one hour, participants and conveners should understand that the meeting might take longer and should make every effort to remain. Special interest groups (e.g. BEU, etc.) will schedule meetings at their own discretion. While the decision to attend any particular meeting shall be left to the individual's professional judgment, such judgment is to be fairly and reasonably exercised. As a matter of normal courtesy, the teacher is expected to notify his/her appropriate supervisor if unable to attend.

- B. Except in an emergency, teachers in grades K-6 do not have to attend curriculum meetings for the first nine workdays of the school year or during June.
- **4.7 Registers of Attendance**. Registers of attendance in all schools shall be done by data processing.
- **4.8 Instructional Aides**. There shall be one building aide at each elementary school (two at the Edward Devotion School).
- **4.9 Teacher Indemnity**. The BSC recognizes its statutory duties and powers under Chapter 258 of the General Laws.

4.10 Health and Safety

- A. A joint committee (JC) consisting of two representatives each designated by the BSC and BEU will be established. The JC shall determine whether additional members shall be added. The JC will review existing health and safety policies and will develop recommendations to the BSC for establishing procedures designed to help identify problems relating to health and safety and to assure prompt investigation and, as appropriate and feasible, corrective action.
- A. The BSC agrees to maintain a healthful and safe work place. Disputes hereunder shall be subject to the grievance procedure but shall not be arbitrable. The parties will utilize a mutually agreed upon private mediator to assist in reaching agreement on unresolved disputes. The cost will be shared equally by both parties.
- **4.11 Class Size Report.** The Assistant Superintendent for Human Resources will provide to the BEU by October 30 and February 28 of each school year a comprehensive report on elementary class size and composition.

Article 5: Leaves of Absence

5.1 Sick Leave

- A. All full-time teachers will be credited with 12 days of sick leave as of the first official day of each school year. Any teacher who joins the staff after the commencement of the school year shall be credited with sick leave proportionately. Any teacher who is on unpaid leave for more than one-half (1/2) of the work year will be credited with sick leave proportional to the time paid. However, if a teacher is on leave for more than one-half (1/2) of the work year for the convenience of the Public Schools of Brookline, s/he shall receive the full sick leave benefit.
- B. Teachers who are employed part-time for 5 days per week shall be entitled to 12 proportionate sick leave days.

- C. A teacher employed for 4 full days per week is entitled to 10 days of sick leave. A teacher employed for 3 full days per week is entitled to 7 days of sick leave. A teacher employed for 2 full days per week is entitled to 5 days of sick leave. A teacher employed for 1 full day per week is entitled to 2 days of sick leave.
- D. Unused days of absence for personal illness will be cumulative each year, without limit.
- E. **Notification Concerning Sick Leave**. In order to allow a reasonable time to provide for a replacement, any teacher who foresees a sick leave of absence of ten or more consecutive days should notify the Assistant Superintendent for Human Resources two months in advance of the approximate date of such absence; but the teacher is encouraged to notify the Assistant Superintendent for Human Resources as soon as possible to allow more time to provide a replacement.
- F. **Family Leave**. It is agreed that the Family and Medical Leave Act ("Act") and the Massachusetts Maternity Leave Act apply to employees in the bargaining unit. FMLA benefits will apply to members of the bargaining unit who are regularly scheduled 0.5 FTE or greater.
- G. The Family and Medical Leave Act (the "Act" or "FMLA")
 - i. FMLA permits unpaid leave to enable employees to care for certain family members with a serious health condition and for the birth or adoption of a child under the terms of the Act. The FMLA "year" shall be a rolling 12-month period. Sick leave taken under this Article shall be credited against any FMLA entitlement.
 - ii. An employee may use up to 12 of his/her earned but unused paid sick leave days per year for a serious health condition of a family member as defined in the FMLA (spouse, child, parent) or of any other relative who is a permanent member of the employee's household or a domestic partner of such employee or for the birth or adoption of a child. If an employee is eligible to use 12 paid FMLA days for the birth of a child, earned but unused sick leave days can be taken subsequent to any sick leave taken for the birth mother's period of disability. Only employees eligible for FMLA, as provided for in Section 5.1.F under this contract, can use the 12 paid FMLA days for any of the above-stated reasons.
 - iii. Effective September 1, 2011, an employee may use up to five of such 12 earned but unused paid sick leave days per year (and/or administrative leave days under Article 5.4A) in case of illness of such family member or other relative which is less than a "serious health condition." If the family member who is ill does not reside in the employee's household, it is expected that such leave will be utilized for the purpose of providing needed care and/or attendance to such person where such care is not otherwise available. All employees, regardless of FMLA eligibility, are eligible for the use of these five (5) days.

H. The provisions of the Massachusetts Workers' Compensation Act shall apply to nurses who are absent due to injuries and/or illnesses covered by that Act to the extent permitted by action of the Town Meeting.

5.2 Sick Leave Bank

A "Sick Leave Bank" shall be maintained whereby all teachers may "deposit" one day of the sick leave to which they are entitled each year. Teachers who have used up the sick leave to which they are entitled may apply to draw on the bank. The bank shall be administered by a Sick Leave Bank Committee consisting of four members, two designated by the BSC and two designated by the BEU. The operation of the bank and withdrawals there from shall be carried out in accordance with the following guidelines:

- A. All deposits to the Sick Leave Bank are to be voluntary.
- B. A teacher shall be enrolled in the Bank unless s/he provides written notice (or via e-mail) of her/his election not to join the Bank to the Assistant Superintendent for Human Resources by October 15th. Such employee's "opt-out" of the Bank shall continue from year to year unless s/he provides written notice (or via e-mail) to join the Bank to the Assistant Superintendent for Human Resources by October 15th of a subsequent year. Any such subsequent joining of the Bank shall be considered the 1st year of membership under section K of Article 5.2.
- C. Any teacher joining the staff after the opening of school will be enrolled in the Bank unless s/he indicates in writing (or via e-mail) to the Human Resources Director within 30 days of initial employment that s/he does not wish to join the Bank
- D. Only those who deposit to the bank may be permitted to withdraw from the bank.
- E. All requests for withdrawal from the bank must be accompanied by the reason and approved by the Sick Leave Bank Committee.
- F. The number of days requested from the bank must be specified at the time of initial request.
- G. Any request for extension of days after the initial request from the bank will be reviewed by the Sick Leave Bank Committee and granted accordingly.
- H. The Sick Leave Bank Committee may request an attending physician's statement regarding the nature of the illness of the applicant and the anticipated period of absence.
- I. The borrower is required to repay 25% of the number of days. This amount may be adjusted by the Sick Leave Bank Committee dependent upon the total days available in the bank.

- J. No days may be withdrawn from the Sick Leave Bank for use by any reason other than illness. Days may not be withdrawn to permit a teacher to stay at home for other members of a family. In the event of a new contract and/or an extension of the existing one, the balance of days in the Sick Leave Bank is to be carried over to succeeding contracts.
- K. During the first year of membership in the Sick Leave Bank, a teacher may not withdraw more than 60 days per school year. During the second year of membership in the Sick Leave Bank, a teacher may not withdraw more than 90 days per school year.
- L. **Subrogation**. An employee who requests sick leave bank days shall be required, as a condition of receipt of such days, to agree in writing to repay the total cost of such days to the Public Schools of Brookline if and when such an employee collects damages from a third party for his/her illness/injury where such illness/injury was the reason provided in accordance with Section E of Article 5.2 and resulted in the need for such sick leave bank days. For any funds recovered, the appropriate number of sick leave days will be restored to the sick leave bank.

5.3 Medical Documentation Protocols

- A. **Health Care Provider's Letter**. If requested by the Assistant Superintendent for Human Resources, the employee shall furnish a letter from a health care provider. Generally, this note is not requested until after the fourth consecutive day of absence. This letter should state the nature of the illness and the anticipated period of absence.
- B. **DOL Form**. If the Superintendent or his/her designee, applying reasonable standards, finds the health care provider's letter to be insufficient, the employee shall present the health care provider's findings as soon as is reasonably possible using the United States Department of Labor's Certification of Health Care Provider form.
- C. **Permission for Consultation**. If the employee submits a complete DOL form signed by the health care provider, the Superintendent or his/her designee may not request additional information from the employee's health care provider. However, the Superintendent or his/her designee, applying reasonable standards, may have a health care provider representing the BSC contact the employee's health care provider, with the employee's permission, for purposes of clarification and authenticity of the medical certification. The employee gives such permission via e-mail or by a hard copy with signature and date. Refusal to give permission will result in no sick leave benefit.
- D. **Second opinion**. If the Superintendent or his/her designee has reason to doubt the validity of a medical certification, the Superintendent or his/her designee may require, in writing, the employee to obtain a second opinion at the BSC's expense. The Superintendent or his/her designee is permitted to designate the health care provider to furnish the second opinion, but the selected health care provider may not be

- employed on a regular basis by the BSC. Any contacts under this paragraph between the Superintendent or his/her designee or his/her health care provider and the designated second opinion health care provider shall be in writing.
- E. **Third opinion**. If the opinions of the employee's and the BSC's designated health care providers differ, the Superintendent or his/her designee may require the employee to obtain certification from a third health care provider, again at the BSC's expense. This third opinion shall be final and binding. The third health care provider must be designated or approved jointly by the Superintendent or his/her designee and the employee must each act in good faith to attempt to reach agreement on whom to select for the third opinion provider. If the Superintendent or his/her designee does not attempt in good faith to reach agreement, the Superintendent or his/her designee will be bound by the first certification. If the employee does not attempt in good faith to reach agreement, the employee will be bound by the second certification. Any contacts under this paragraph between the Superintendent or his/her designee or its health care provider and the designated third opinion health care provider shall be in writing.

F. Additional rules for 2nd and 3rd opinions

- i. Pending receipt of the second or third medical opinion, the employee is provisionally entitled to the sick time. If the certifications do not ultimately establish the employee's entitlement to the sick time, the sick leave shall be treated as unpaid leave and the employee shall be required to repay any pay received during the period of provisional entitlement.
- ii. If the Superintendent or his/her designee requires the employee to obtain either a second or third opinion the BSC must reimburse the employee for any reasonable "out of pocket" travel expenses incurred to obtain the second and third medical opinions. The Superintendent or his/her designee may not require the employee to travel outside normal commuting distance for purposes of obtaining the second or third medical opinions except in very unusual circumstances.
- iii. Copies of 2nd and 3rd opinions received by the Superintendent or his/her designee are to be mailed to the employee within two business days of the receipt by the Superintendent or his/her designee.
- G. **Files of Medical Documentation**. Any and all medical information provided by an employee and/or his/her medical care professional and/or any medical care professional contracted by the BSC (or by the BSC and the employee jointly) shall be held as strictly confidential and placed in a medical file separate from the employee's personnel files. Physical access to these files shall be limited to the employee, the Superintendent and the Assistant Superintendent for Human Resources and their respective confidential secretaries. Both parties reserve all rights under law with respect to privacy of medical information.

5.4 Requests for Leaves, Extensions or Renewals. All requests for leaves, extensions or renewals of leaves shall be made in writing (or via e-mail) and responded to in writing. Requests for such leaves and renewals shall be submitted to the Superintendent or his/her designee with a copy to the teacher's immediate superior by February 1 prior to the commencement of the leave, whenever possible. If such submission is delayed to not later than March 15, the Superintendent shall approve such request if the delayed submission is due to extenuating circumstances of the individual applicant as reasonably determined by the Superintendent. Notwithstanding any provisions of Section 5.6 (A), (B), (C in case of enlistment only), (F) or (H) of this Agreement to the contrary but subject to applicable laws, the Superintendent shall not be required to grant a leave of absence requested after July 31.

5.5 Leaves of Absence with Pay

A. Administrative Leave. Each teacher shall be entitled to a total of four days of leave with pay during each school year. The teacher is required, with respect to two of such four days of leave with pay to give no reason to any person, but is required with respect to the other two such days of leave to state orally, to the teacher's immediate supervisor the reason for taking such leave, provided that if any of such days falls on a Monday, a Friday or on a school day immediately preceding or following a holiday, a reason must be stated. Each nurse shall be entitled to five (5) days of administrative leave with pay during the school year; no more than two (2) of said days may be taken without the advance approval of the Superintendent or his/her designee. The parties agree that such leave should be taken only for important personal reasons and not as vacations or for recreation. Except in emergencies, each person proposing to take leave pursuant to this section shall so notify the Superintendent or his/her designee at least five school days before the date on which the leave is to be taken or to commence. The Superintendent or his/her designee, if he/she determines that such person's absence in accordance with the notice would seriously jeopardize the normal functioning of the school system, may require a statement of reasons and upon hearing them may instruct the person not to take the leave at the time proposed. Any employee may be required to meet with his/her supervisor(s) regarding a pattern of use of administrative leave days and/or potential misuse of administrative leave days. The supervisor will address any determination of misuse through appropriate discipline.

Any teacher who is on unpaid leave for more than one-half (1/2) of the work year will be credited with administrative leave proportional to the time paid. However, if a teacher is on leave for more than one-half (1/2) of a work year for the convenience of the Public Schools of Brookline, s/he shall receive the full benefit.

If at the end of any work year a teacher has unused Administrative Leave days, such days shall be converted to sick leave.

B. Each teacher shall be granted one (1) day per year to visit other schools or to attend activities which have a demonstrable relationship to the improvement of his/her professional skills and expertise.

The Superintendent shall have the authority to grant more than one day to a particular person, if s/he is convinced such leave will be of direct benefit to the Public Schools of Brookline.

C. Bereavement

- i. Teachers shall be granted leave of absence with pay for not more than five days (not necessarily in Succession) on account of the death of father, mother, brother, sister, husband, wife, child, son-in-law, daughter-in-law, or parent-in-law, whether such relative was a member of the teacher's household or not. Pay for absence not to exceed five (5) days shall also be allowed on account of death of any other relative who was a permanent member of the teacher's household, or another person with whom said teacher resided.
- ii. Teachers shall also be paid full salary for an absence not to exceed one day to attend the funeral of a first cousin, grandparent, grandchild, brother-in-law, sister-in-law, aunt, uncle, nephew, or niece.
- iii. For purposes of this section C, domestic partner shall be treated on the same basis as a spouse.
- D. No salary deduction shall be made in case of absence by reason of being called to appear in court or other tribunal on business concerning the Public Schools of Brookline.
- E. Each member of the professional staff covered by this Contract who is a member of a state or national reserve complement shall be entitled to up to ten (10) days each school year to serve with said reserve unit. The BSC agrees to pay the differential between military pay and the teacher's salary when the former is smaller, during this period of temporary active duty. Every effort shall be made by a teacher in cooperation with the Superintendent to arrange for the fulfilling of active duty reserve obligation at times other than when schools are in session.
- F. If a teacher's attendance at summer school is required by the Superintendent or his/her designee and if said summer school begins prior to the end of the teacher's work year, he/she shall be released in sufficient time to commence attending the summer school according to schedule without loss of pay.
- G. **Jury Duty**. No teacher shall suffer a loss of salary to which he/she would otherwise be entitled for performing jury duty. The Assistant Superintendent for Human Resources will request exemption from jury duty in accordance with applicable laws. Teachers who are not excused from jury duty must notify the Assistant Superintendent for Human Resources and exercise any rights they have to seek to reschedule their jury duty service at a time not in conflict with the school calendar. If a teacher complies with this section but is required to perform jury duty service on days he/she otherwise would have worked, such days shall not be charged to any other type of leave.

H. Work-Related Injury or Illness. An employee who is absent due to a work-related injury or specified work-related illness shall submit a copy of the accident report and supporting medical documentation to the Superintendent or his/her designee who will investigate the incident and present the information to the Sick Leave Bank Committee. The Sick Leave Bank Committee shall advise the Superintendent or his/her designee regarding coverage for such absence without loss of pay or sick leave according to the following guidelines of the BSC. The decision of the Superintendent or his/her designee is final and is not subject to the grievance procedure.

Providing that an employee can establish that the illness is work related, the following illnesses shall be covered: measles, mumps, chicken pox, fifths disease, pertussis and meningitis. All other illnesses and diseases, including mental illness, emotional distress, and illness caused by stress-related conditions, shall be excluded.

Teachers may be reimbursed for medical expenses (excluding co-payments and deductibles) not covered by insurance for work-related injuries and covered work-related illnesses up to a maximum of \$250 per teacher per year subject to a \$5,500 cap on the aggregate liability of the BSC per year for all members of Unit A and Unit B. In the event a teacher has reimbursable medical expenses which exceed \$250 and the BSC has not exhausted the annual \$5,500 cap, the teacher may request additional reimbursement from the Superintendent or his/her designee.

5.6 Leaves of Absence without Pay

A. Union Leaves

i. Unpaid Leave for Certain BEU Officers. With the approval of the Superintendent, up to three (3) members of the professional staff (comprised of both Units A and B), designated by the BEU shall, upon request, be granted leaves of absence for not more than two (2) years without pay for the purposes of serving as elected officers of the BEU at the state or national level. Upon return from such leave, a teacher shall be considered as if he/she were actively employed by the BSC during the leave and shall be placed on the salary schedule at the level he/she would have achieved, had he or she not been absent.

ii. Paid Leave for BEU President

a. A full time paid leave of absence shall be granted annually to the BEU President in each year of this Agreement for the purpose of conducting BEU business. The BEU agrees to reimburse the BSC for the entire cost of wages and all fringe benefits attributable to each paid leave. The reimbursement of wages will be calculated by adding the President's salary for that year to the average salary of persons newly employed on the same salary schedule as the President on or before September 1 of that year, and by dividing such total by two. Such reimbursement shall in no event be higher than the BEU

- President's salary. In any year when there are no new hires, the parties will use the average salary calculated from the prior year, adjusted by the percentage change on the salary schedule for the year of no new hires.
- b. The BEU must notify the Superintendent by March 1 of the intention of the BEU President not to return to active employment in his or her bargaining unit the following school year.
- c. The classroom re-entry procedures applicable to the BEU President in office on March 1, 2000 are set forth in a side letter dated March 22, 2000. Classroom re-entry procedures applicable to a subsequent Union President who has been absent from a classroom for more than four consecutive years while on Union President leave will be subject to negotiations between the parties in a successor agreement.
- B. A leave of absence without pay of up to two (2) years shall be granted to any teacher who joins the Peace Corps, VISTA, Massachusetts Teacher Corps, or serves as an exchange teacher, and is a full-time participant in any such program. Upon return from such leave, a teacher will be considered as if he/she were actively employed by the BSC during the leave and will be placed on the salary schedule at the level he/she would have achieved had he/she not been absent.
- C. Leaves without pay shall be granted to any teacher who is inducted or enlists in any branch of the armed forces of the United States. Upon return from such leave, a teacher shall be placed on the salary schedule at the level which he/she would have achieved had he/she remained actively employed in the system during the period of his/her absence, up to a maximum of four years.
- D. A leave of absence without pay or increment of up to one (1) year shall be granted for the purpose of caring for a sick member of the teacher's immediate family. Additional leave may be granted at the discretion of the Superintendent or his/her designee.
- E. After four (4) years continuous employment in the Brookline School System, a teacher may be granted a leave of absence without pay, for up to one (1) year for health reasons. Requests for such leave will be supported by appropriate medical evidence. Any teacher whose personal illness extends beyond the period compensated shall be granted such a leave of absence.
- F. A leave of absence of up to one (1) year shall be granted to any teacher not eligible for Sabbatical Leave as provided for in Section 5.10.B of this Contract for purposes of engaging in study or for cultural travel or work programs related to his/her professional responsibilities Upon return from such work, a teacher shall be placed on the salary schedule at the level he/she would have achieved had he/she remained actively employed in the system during the period of his/her absence.

- G. All benefits to which a teacher was entitled at the time his/her leave of absence commenced, including unused accumulated sick leave shall be restored to him/her upon his/her return, and he/she shall be assigned to a position equivalent to the one which he/she held at the time said leave commenced.
- H. Other leaves of absence without pay may be granted by the Superintendent or his/her designee.
- **Superintendent's Approval of Leaves**. Notwithstanding the foregoing, it is agreed that a leave of absence which is to begin or end during the school year must be approved by the Superintendent or his/her designee. Such approval shall be granted only in extraordinary circumstances. Subject to the foregoing, such approval shall not be unreasonably withheld.

5.8 Child-bearing Leave

- A. Employees shall be granted leaves of absence for reasons of childbearing in accordance with applicable federal and state laws.
- B. A teacher on child-bearing leave may apply unused sick leave only for those days on which she is actually incapacitated and unable to work due to disability resulting from pregnancy or recovery there from, except as provided for in Article 5.1.G.ii.

C. Summer births

- i. Sick Days. An employee granted child-rearing leave is paid via use of sick days only when the employee is actually incapacitated and unable to work due to disability resulting from pregnancy, birth and/or recovery there from during the employee's usual days of work. For example, if the employee is incapacitated during the summer, the employee receives no pay. However, for a period of incapacity resulting from pregnancy, birth and/or recovery there from that extends into the work year (e.g. a delivery date of June 2 or August 25), the employee will be paid via use of sick days, even if the employee takes the rest of the school year off on child-rearing leave
- ii. Paid Family Leave. If an employee is eligible to use 12 paid FMLA days for the birth of a child pursuant to Article 5.1.G.ii, the employee is entitled to the 12 paid days of FMLA Leave even if the period of incapacity related to the birth does not extend into the school year and/or the employee is to take the rest of the year off.
- iii. Sick Time, Longevity, Seniority Accrual. An employee does not accrue sick time, seniority, or longevity, for any year in which the employee is paid solely on sick time or the 12 FMLA days, i.e. the employee actually has to work sometime during the year to accrue sick time, seniority, or longevity.

5.9 Child-rearing Leave

A. Entitlement. Any teacher with professional status or nurse with three years of continuous service will be granted a leave of absence without pay for up to two years for the purpose of childrearing. It is understood that teachers on leave will return on September 1 unless prior agreement has been made with the Superintendent or his/her designee. This also applies to adoptive parents. If an employee eligible for childrearing leave gives birth or adopts a child before January 15th of any school year and takes the rest of the school year off, that shall constitute one of years entitled. If such an employee gives birth or adopts a child on or after January 15th of any school year and takes the rest of the school year off for child rearing leave, then the employee shall be entitled to two additional school years off for child rearing leave related to the birth or adoption.

B. 5-month notice

- i. All requests for child-rearing leaves shall be made in writing to the Assistant Superintendent for Human Resources no later than five calendar months before the expected delivery date.
- ii. In the case of adoption, the employee is to request the leave once the employee's home study has been completed and notarized.
- iii. Employees wishing to take a child-rearing leave that starts the school year after the birth year shall give notice by February 1.
- C. Adverse Circumstances. In the case of the termination of the pregnancy for any reason or the death of a baby during a child-bearing or child-rearing leave of absence, the Superintendent or his/her designee will grant the request of the employee to return to work earlier than anticipated.
- D. Superintendent's Discretion. The granting of a child-rearing leave is not subject to the discretion of the Superintendent or his/her designee (i.e. it will be granted automatically) given:
 - i. the leave commences in September or directly after other adoption/birth-related leaves; and
 - ii. the employee returns to work the following September; and
 - iii. the employee is a teacher with professional status; a nurse, occupational therapist or physical therapist with three years of continuous service; and
 - iv. the leave is for the employee's entire position (i.e. not part-time); and
 - v. submission deadlines are met.

All other child-rearing requests may be granted at the reasonable discretion of the Superintendent or his/her designee.

E. Child-bearing and Child-rearing Forms. Employees are to apply for child-bearing and child-rearing leave on the form developed by the BEU and BSC updated March 11, 2004.

5.10 Sabbatical Leave Policy

A. Policy

The policy of sabbatical leave is a plan intended to help maintain and improve the quality and efficiency of instructional service in the Public Schools of Brookline. It is an additional opportunity for teachers to extend their professional competence and so better serve the Brookline Schools. Such growth is particularly vital in a period of rapid changes in the techniques and the art of teaching.

The Superintendent of Schools will recommend to the BSC the granting of such leave as he/she judges will enlarge the professional competence of the applicant and improve the general efficiency of teaching services.

All professional employees are eligible for sabbatical leave and are considered in the following regulations in the term "teachers."

B. Eligibility

- i. Any teacher who has completed seven or more years of continuous satisfactory service in the Public Schools of Brookline may, upon recommendation of the Superintendent, be granted a leave of absence for one semester at full pay or two semesters at half pay for study or research on a full-time basis, subject to Section 5.10.L.i.
- ii. The applicant shall not be eligible for subsequent leaves until he or she has served another period of seven continuous years of successful service.
- C. **Number of Leaves Authorized**. Not more than one percent of all professional staff (in both units A and B combined) shall be granted a sabbatical leave during any given academic year. Requests for sabbatical of less than a full year shall be granted only upon showing of extraordinary circumstances. The BSC reserves the right to exceed the 1% limit for extraordinary circumstances as determined by the BSC.

D. Application for Leave

i. Application shall be made on or before March 1 of any year. If approved, such leave shall officially begin at the beginning of either the first or second semester of the year immediately following. Each applicant shall be notified as to the disposition of his/her application no later than April 30.

ii. Applications shall be made upon a regular blank form prescribed by the Superintendent and shall include a program of study or research to be followed by the teacher during the period of leave.

E. Selection Procedure

- i. In recommending sabbatical leaves of absence, the Superintendent shall give consideration to the use to be made of the requested leave and to seniority in service. Similarly, consideration shall be given to the reasonable and equitable distribution of the applications among the different schools, grade levels, departments and administrative staff.
- ii. Personnel to be granted sabbatical leave shall be selected by a standing committee of three members consisting of the Superintendent of Schools, who shall act as Chairman, the Assistant Superintendent for Human Resources, and the President of the BEU or his/her designee.
- iii. If the number of the professional staff in Units A and B combined who apply for sabbatical leave in any year exceeds one percent, then selections of teachers for sabbatical leaves will be made from the following groups.
 - a. Professional staff with extremely long periods of service without sabbatical leave who require retraining in order to serve effectively.
 - b. Professional staff who wish to upgrade skills in accordance with a plan for improved teaching efficiency or improved classroom practices.
 - c. Professional staff who will undertake new or revised duties in accordance with a request from the administrative staff or the BSC.
 - d. Professional staff who are engaged in a significant research study which will benefit the Brookline Schools or education in general.
 - e. Professional staff whose sabbatical study will lead to an advanced degree.
- F. Subsequent Service. Prior to granting such leave, teachers shall enter into a written agreement with the BSC that upon termination of such leave he/she will return to service in the Public Schools of Brookline for a period equal to twice the length of such leave and that, in default of completing such service, he/she will refund to the Town of Brookline an amount equal to such proportion of salary received by him/her while on leave as the amount of service agreed to be rendered, unless such teacher is incapacitated, has been discharged or has been released for good and sufficient reasons by the BSC from this obligation.
- G. **Professional Teacher Status and Pension Status**. Contributions to the retirement fund by the teacher shall continue through the period of sabbatical leave and shall be credited for service in accordance with the practices of the Massachusetts Teachers

Retirement System. Professional teacher status privileges and salary increments shall continue just as they might in normal employment.

- H. **Illness or Accident**. Should the program of study or research being pursued by the teacher on sabbatical leave be interrupted by serious accident or illness (verified by the Superintendent) this unforeseen fact shall not be considered as a breach of the contractual agreement nor prejudice the teacher against receiving all rights and benefits provided for under the terms of the sabbatical leave policy, providing the Superintendent was notified of such accident or illness by registered letter within 15 days of its occurrence.
- I. **Forfeiture of Leave**. If the teacher on sabbatical leave is not fulfilling the purpose for which the leave was granted in the opinion of the Superintendent, the Superintendent shall report this fact to the BSC. The BSC may terminate the leave of absence as of the date of its abuse as determined by the Superintendent. In the event such leave is terminated, the teacher will return to active duty as assigned by the Superintendent.
- J. **Return to Active Duty**. A teacher on sabbatical leave for the first semester of any school year shall notify the Superintendent in writing on or before December 1 of that year of his/her intention to return to duty the following semester. If leave has been granted for the second semester or the entire school year, the teacher shall give such notification on or before April 1.

K. Reinstatement

- i. Upon the termination of the sabbatical leave, the teacher who has complied with the regulations set in the aforementioned paragraph shall be reinstated in the position he/she held at the time such leave was granted or an equivalent position, unless he/she shall agree otherwise.
- ii. The teacher shall, at the expiration of his sabbatical leave present satisfactory evidence that the period of leave has been utilized in good faith for the purpose of which it was granted before the aforementioned reinstatement is exercised.

L. Salary

- i. The salary granted a teacher on sabbatical leave for one semester shall be the same salary to which he or she would be entitled, if not on leave, minus the regular deductions for Federal Income Tax, Massachusetts Income Tax and for Teachers' Retirement as computed for all employees. The salary granted to a teacher on a full school year sabbatical leave shall be one-half of the salary to which he/she would be entitled if not on leave. The salary granted to a teacher on a full year sabbatical who is serving his/her tenth year or more shall be paid three-fourths of the salary to which he/she would be entitled if not on leave.
- ii. The salary payments shall be in accordance with the general time schedule for payment of salaries in the Public Schools of Brookline.

M. **Recommendations by the BEU**. From time to time the BEU may make recommendations to the Superintendent concerning the provisions of the Sabbatical Leave Policy.

Article 6: Evaluation and Supervision

6.1 Evaluation and Supervision

- A. The Annual Performance Evaluation Report will be based upon the Brookline Principles of Effective Performance and Examples of Descriptors (found in the Evaluation Principals and Format Appendix under separate cover, incorporated into this Agreement and, as such, enforceable under Article 9) and the Alternative Evaluation Process: Professional Growth Project document included as Appendix E in this Agreement. The procedure for evaluation shall be as set forth in Article 6.2.
- B. Annual Reports and Progress Reports, as described in Article 6.2 and found in the Evaluation Principals and Format Appendix under separate cover, must be submitted by principals in the Elementary Schools and by secondary curriculum coordinators in the High School (in cases where there is no secondary curriculum coordinator, by the Headmaster). Annual Reports and Progress Reports may be submitted by Curriculum Coordinators, Deans, Headmaster and/or Assistant Superintendent for Human Resources.

Occupational therapists and physical therapists shall be evaluated by Special Education and/or Early Childhood Coordinators unless assigned to a building administrator or senior administrator by the Superintendent.

Deans will cooperate in requested evaluations of pre-professional teachers in accordance with practices followed by housemasters in the past.

Upon the request of the headmaster, a dean will conduct an evaluation of a teacher with professional status following the procedures established in the contract. Absent unforeseen circumstances, the headmaster will attempt to notify the deans about such teachers at the beginning of the school year.

C. The BSC and the BEU endorse the concept of student/teacher interaction and encourage teachers actively to seek and use student evaluation. All teachers may be required to administer the student evaluation form adopted by the BSC. The completed forms may be reviewed by a teacher with his/her supervisor at either's request provided that a teacher shall not be required to review or share the completed form with his/her supervisor.

A committee consisting of an equal number of representatives of the BSC and the BEU, may be convened during the terms of this contract at the request of the BSC, to study the issue of student evaluation of teachers and to make recommendations to the BSC relating to the process of student evaluation. The student evaluation process described herein shall not be changed during the term of the agreement unless otherwise mutually agreed by the BSC and the BEU.

- D. The BSC encourages continual peer observation-supervision at all levels for the purpose of mutual sharing, developing and examining of teacher styles, approaches and techniques.
- E. When it becomes necessary to engage teachers in the formal evaluative process (by request from evaluators or the teacher being evaluated), then the following formula should be used:
 - i. In the High School, the consensus of the teacher's department shall determine the selection of one peer evaluator, and the consensus of the high school shall determine the selection of additional peer evaluators.
 - ii. In the Elementary Schools, the teacher involved in peer evaluation will be determined by a consensus of the building faculty; or in the case of seventh and eighth grade specialist, by a consensus of the Town-Wide Department.
 - iii. In both cases the number of peer evaluators and administrators shall be equal.
 - iv. For teachers with Professional Status, supervisors and evaluators always have the possibility of initiating a request for another evaluator (administrator or peer).
- F. The BSC encourages self-evaluation by teachers with pre-professional status and teachers with professional status. The mechanism for this might be either a checklist filled out by the teacher or an Annual Performance Evaluation Report completed by the teacher. Self evaluation can be requested by either the teacher or the supervisor. It may be followed up by conference if it is desired by either party.
- G. It is strongly suggested that supervisors provide for visitations by teachers within departments, among departments, and among buildings and other school systems. Also, supervisors are encouraged to provide in-service workshops for teachers.

6.2 Procedure for Evaluation

A. **Frequency of Evaluation**. Teachers with pre-professional status will be evaluated every year. Upon achieving professional status (PTS), teachers will be evaluated on a four (4) year evaluation cycle, effective for the 2010-2011 school year.

The Professional Teacher Status (PTS) Evaluation Cycle:

- Year 1 No formal evaluation process
- Year 2 Comprehensive Evaluation **OR** Professional Growth Project
- Year 3 No formal process
- Year 4 Comprehensive Evaluation **OR** Professional Growth Project

The Professional Growth Project (PGP) is intended to offer PTS educators the opportunity to engage in reflection, inquiry, and analysis of their teaching and develop new skills and knowledge about their teaching and student learning. Sharing

the goal of improving student learning by improving teaching, the PGP is part of the evaluation process as an alternative to the Comprehensive Evaluation for educators with PTS.

For Year 2 and Year 4, the educator may request either the Comprehensive Evaluation or Professional Growth Project. The educator's supervisor (Principal, Headmaster, or BHS Coordinator) may accept the educator's request for a PGP, or the supervisor may deny the request and require the educator to undergo the Comprehensive Evaluation. This decision is entirely the supervisor's. The supervisor is required to give specific reasons if s/he denies the PGP option to an educator.

If an educator requests participation in a PGP, and is approved by the supervisor, the next formal evaluation, two (2) years later, will be the Comprehensive Evaluation. In other words, an educator is to receive a Comprehensive Evaluation at least once within a four (4) year period.

Within the four year cycle at elementary schools, the educator's Principal or Vice Principal will be the evaluator, at least once.

For the remainder of Article 6.2, references to teachers with professional status refer only to their year of evaluation.

- B. **Orientation**. In September of each year, all teachers with professional status and teachers with pre-professional status will meet with supervisors for an orientation of the evaluation procedures. It is the intent of this meeting to allow all staff the opportunity to enter the evaluation process with full and open knowledge of what the process is.
- C. **Pre-Evaluation Conference**. On or before October 15, supervisors shall meet with all teachers with pre-professional status. On or before December 1, supervisors shall meet with all teachers with professional status. These pre-evaluation conferences will address goals for the year, strengths of the teacher, relationships between current and previous evaluations and support available.

D. Classroom Observations

- i. There will be one announced, formal classroom observation for each teacher with pre-professional status and for each teacher with professional status. Each formal classroom observation will be at least thirty (30) minutes in length for elementary teachers and a full period for secondary teachers.
- ii. Supervisors may informally observe teachers at any time.
- E. **Pre-Observation Conference**. A pre-observation conference may take place prior to a formal classroom observation. Requests for these conferences may be made by either the supervisor or the teacher.

- F. **Post-Observation Conference**. A post-observation conference shall be held within seven (7) work days of the classroom observation, unless the supervisor and the teacher agree to waive the seven (7) day time line and arrange another mutually convenient date. A classroom observation report shall be completed at or after the post-observation conference. A copy of the completed observation form, signed by the teacher and the supervisor, will be given to the teacher.
- G. **Staff Progress Report**. Each teacher with pre-professional status shall receive a Staff Progress Report (found in the Evaluation Principals and Format Appendix) three times a year, normally every two months. For teachers with professional status, Staff Progress Reports may be used if desired.
- H. **Annual Performance Evaluation Reports**. An Annual Performance Evaluation Report (found in the Evaluation Principals and Format Appendix) will be made prior to February 15 each year for every teacher with pre-professional status except that the Superintendent can extend the period to a date not later than March 15 when, in his/her judgment, there is good reason for doing so. An Annual Performance Evaluation Reports will be made prior to May 1 of each year for every teacher with professional status.

Each teacher shall receive a copy of his/her report and have the right to discuss the report with his/her superior and submit his/her own comments within ten days. It is expected that the evaluator will have conferred periodically with the teacher concerning his/her progress before the submission of the Annual Performance Evaluation Reports.

- I. Non-renewal of teachers with pre-professional status. A teacher with pre-professional status shall be notified, in writing, on or before April 15 if he/she is not to be employed for the following school year. If, however, the teacher is to be reinstated, s/he shall be duly informed of such action in writing by the administrator making the decision or the Assistant Superintendent for Human Resources, whichever is appropriate.
- J. Out-Of-Cycle Evaluation. Any teacher with professional status may be evaluated out-of-cycle. On or before October 20, the supervisor will provide the teacher with the substantive reasons for undertaking the out-of-cycle evaluation, and it is understood this action does not necessarily imply a negative evaluative finding. This decision to undertake an evaluation out of cycle, though grieveable, shall not be subject to arbitration.

When the performance of a teacher with professional status is considered satisfactory after having been evaluated out-of-cycle, that teacher's cycle will begin anew at the time of the satisfactory evaluation.

6.3 Duration of Evaluation System. For the duration of this Agreement, i.e., through August 31, 2011, present system-wide practices concerning evaluations will continue in effect, with the Alternative Evaluation Process being implemented beginning in the second year of the Agreement (2010-2011).

- **6.4 Withholding Salary Increment**. Superintendent reserves the right to withhold a salary increment from any teacher who has not performed in a satisfactory manner. If a teacher is at the maximum of his/her preparation column and has not performed in a satisfactory manner, the Superintendent reserves the right to withhold any salary increases.
- **6.5 Personnel File.** Each teacher shall have the right to review and make copies of the contents of his/her personnel file originating after initial employment and to have a representative of the BEU accompany him/her in such review.
- 6.6 Derogatory Material. Should any material which the Superintendent or his/her designee shall consider to be derogatory be placed in the teacher's personnel folder, then the teacher involved will be given the opportunity to review same prior to its placement in the personnel folder. The employee shall initial such material prior to its insertion. The employee's initials do not signify agreement with the contents but rather serve as evidence that the employee has seen the material in question. If the teacher chooses to do so, he/she may submit any statement concerning the derogatory material; and if he/she so desires, said statement shall be filed with the alleged derogatory material in the teacher's personnel folder. If such material may result in discipline or otherwise affect the

teacher's status, it shall be sent to the teacher involved within twenty (20) school days. It is understood that the Superintendent or his/her designee will make an appropriate investigation before inserting derogatory material into an employee's file. In addition, with regard to nurses, any complaint to any school official regarding the performance of a nurse which the school official believes could affect the nurse's employment status, shall be called to the attention of the nurse involved within 7 working days.

6.7 Academic Freedom

- A. Every teacher shall have the right to active citizenship, to participate in any political activity or to run for public office.
- B. The teacher may, with the approval of the Principal or Secondary Curriculum Coordinator, introduce additional materials to be used or presented in the classroom.

6.8 Professional Development and Educational Improvement

- A. The BSC shall pay reasonable expenses, in accordance with normal practice, for a teacher requested to attend a workshop or seminar.
- B. Vouchers for courses shall be given preferentially to the cooperating teachers who have earned them. Upon request of the BEU, the Assistant Superintendent for Human Resources and a representative of the BEU will review the availability and distribution of vouchers in December and May of each school year.

- C. **Professional Development Council.** The Superintendent, through the Deputy Superintendent for Teaching and Learning, will establish a Professional Development Council for the purpose of addressing the professional development needs of all bargaining unit members. At a minimum, the Council shall consist of the following: one preschool-grade 2 teacher, one grades 3-6 teacher, one grades 7-8 teacher, two grades 9-12 teachers, one elementary special education teacher, one high school special education teacher, one teaching specialist (e.g. art, music, PE), one non-classroom specialist (e.g. social worker, guidance counselor), one member from Unit B, and one paraprofessional. The BEU shall appoint one member of the Council. The Deputy Superintendent, after consultation with the BEU, will appoint the remaining members of the Council.
- D. **Joint Labor Management Committee**. The parties shall convene a Joint Labor Management Committee (JLMC) to make recommendations regarding the negotiable elements of educator evaluation in compliance with the 2011 revised regulations of the Massachusetts Department of Elementary and Secondary Education. In addition, the JLMC shall make recommendations regarding the role of Unit A members in the evaluation of paraprofessionals. The JLMC shall consist of three (3) Unit A representatives appointed by the BEU and three (3) management members appointed by the Superintendent. The JLMC shall submit recommendations on negotiable elements of educator evaluation and on the role of Unit A members in evaluating paraprofessionals to the School Committee and BEU no later than June 30, 2012 with the goal of piloting some or all elements of the new evaluation system in the 2012-2013 work year followed by full implementation of the new evaluation system in the 2013-2014 work year.
- **Employee Assistance Program**. Members of this bargaining unit and their household members will have access to a confidential employee assistance program that provides assessment and referral services as well as short-term counseling.
- **6.10 Evaluation of Administrators**. Staff members are encouraged to provide feedback directly to supervisors. However, any staff member may submit a report regarding his/her supervisor at any time during the school year. This report shall be signed by the sender and submitted to the Assistant Superintendent for Human Resources and to the concerned supervisor.

6.11 Discipline with Discretion

- A. An employee will be notified, in advance, in writing (or via e-mail) when practicable, of the purpose of a meeting with an administrator in cases where disciplinary action is contemplated, and shall be entitled to have BEU representation.
- B. The administrator shall exercise reasonable discretion in the manner in which s/he disciplines an employee. When an administrator criticizes or disciplines an employee, those discussions shall not occur in the presence of parents, students or other employees not superior to the employee.

Article 7: Just Cause

No teacher who has acquired professional teacher status shall be discharged or otherwise disciplined without just cause.

No grievance involving the dismissal, demotion or suspension of a teacher shall be submitted to arbitration unless the teacher by said choice waives any rights he or she may have under Chapter 71 of the General Laws of the Commonwealth of Massachusetts to resolve the same dispute.

This provision shall not apply to the non-renewal of the annual contract of persons holding positions listed in Appendix B-8.

Article 8: Conditions of Development and Implementation of Educational Programs

- **8.1 Policy Recommendations**. The BEU and the BSC desire to encourage the active participation of the teachers in the development and implementation of the best possible programs for the children of Brookline, including State and Federal Aid Programs. The Faculty Representative Council as the standing committee of the BEU may make recommendations to the Superintendent in regard to such programs. The Superintendent or his/her designee shall acknowledge and review such recommendations and meet with the Faculty Representative Council to discuss them.
- **8.2 Policy Review**. Before a new policy is recommended to the BSC, it shall be presented to the Faculty Representative Council for their response within a week thereafter.
- **8.3 In-Service Training**. Teachers who wish to make suggestions for in-service training programs are encouraged to do so by submitting any such suggestions to a Teachers and Administrators' Training Fund (TATF) subcommittee consisting of two representatives from Units A and B combined designated by the BEU and two designated by the BSC.

Article 9: Grievance Procedure

- **9.1 Definition of Grievance**. Any claim by the BEU or a teacher that there has been a violation, misinterpretation or misapplication of the terms of this contract or a violation of any established policy or practice shall be a grievance. As used in this section the words "established policy" shall mean a policy which is established by the BSC and which may be changed by the BSC from time to time pursuant to the powers conferred upon the BSC by law.
- **9.2 Time Limits**. During the school year the time limits specified in this article shall mean school days. During the summer recess the time limits shall mean calendar days exclusive of Saturdays, Sundays and legal holidays. The time limits specified in this article shall be considered maximal unless extended by mutual agreement in writing.

- **9.3 Level One.** A teacher with a grievance shall present it within thirty (30) days after the occurrence giving rise to the grievance or after the teacher, using due diligence, should have become aware of such occurrence, to his/her Principal, the Headmaster or his/her immediate supervisor, in writing on a standard form, either directly or through the BEU. If the occurrence which gives rise to the grievance was the result of a decision or an action of the Superintendent, the Level One grievance shall be presented to him/her.
- **9.4 Level Two**. If the grievance is not resolved to the satisfaction of the grievant within five (5) days after submission at Level One, the employee may present the written grievance to the BEU's Grievance Committee within five (5) days thereafter. The Grievance Committee shall review the grievance with the aggrieved and give its decision to the employee within five (5) days after receipt of the grievance. The BEU Grievance Committee may present the grievance in writing to the Superintendent within five (5) days thereafter.
- **9.5 Level Three**. If the grievance is not resolved to the satisfaction of the grievant or the BEU within five (5) days after submission at Level Two, the teacher or the BEU may present the grievance in writing to the BSC within five (5) days thereafter.
- **Level Four**. If the grievance is not resolved to the satisfaction of the BEU within fifteen (15) days after submission at Level Three, the BEU may submit the grievance to binding arbitration under the Voluntary Labor Arbitration Rules of the American Arbitration Association within twenty (20) days thereafter.
- 9.7 Resolving Grievances. If (a) a grievance is submitted at Level One, (b) it is not resolved to the satisfaction of the grievant or the BEU within five (5) days after such submission, and (c) at the end of the said five (5) days there are less than fifty (50) days remaining before the expiration of this contract, then the grievant or the BEU may present the grievance in writing to the BSC at Level Three within ten (10) days thereafter. No grievance originally presented during the term of this contract shall lapse by reason of the termination of this contract, unless such a grievance could not be a grievance under the successor contract. For the purposes of resolving any grievance to which the preceding sentence applies, the terms of this contract shall be deemed to remain in effect until such grievance is finally resolved. The foregoing sentence shall not be construed as extending the term of this contract for any other purpose or any other circumstances.
- **9.8 Arbitrability**. BSC decisions on any question of educational policy or any decision regarding contract renewal, reappointment or dismissal of a teacher with pre-professional status shall not be subject to arbitration. No Arbitrator shall have any power to alter, add to or detract from the provisions of this Agreement.

9.9 General Provisions

- A. The BEU shall have the right to use in its presentation at any level of this grievance procedure any representative(s) of its own choosing.
- B. The expenses of any arbitration under this Article shall be borne in accordance with the Voluntary Labor Arbitration Rules of the American Arbitration Association.

- C. All decisions at Level Two and Three of the Grievance Procedure shall be in writing and shall be promptly presented to the grievant or the BEU.
- D. The attendance of any teacher at a grievance hearing, whether as the grievant, as a person asked to furnish or verify facts or as a representative of the BEU, shall be regarded as a professional responsibility to be performed without loss of pay.
- E. The parties agree to make reasonable attempts to resolve grievances informally before they are presented at Level One.
- **9.10** Legal and Equitable Remedies. Nothing in this Article shall prevent either party from instituting any proceeding to obtain any legal or equitable remedy to which it may be entitled before any court or other tribunal of competent jurisdiction.

Article 10: BEU Rights and Privileges

10.1 Use of Buildings and Facilities

- A. The BEU shall have the right to use the school buildings at reasonable times for meetings. If the meeting is held during those hours when a custodian is present or has regularly scheduled hours, there will be no charge for the use of the building.
- B. The BEU shall have the right to post materials dealing with proper and legitimate business of the BEU on bulletin boards used for teachers' exchange.
- C. The BEU shall have the right to use the teachers' mailboxes for distribution of materials.
- D. The BEU President can continue to have an e-mail account the use of which must comply with all applicable laws and regulations.
- E. The BSC shall continue to make office space, comparable to present size, available to the BEU.
- **10.2 Right to Address**. The BEU, like all citizens, has the right to speak to the BSC on matters of mutual concern during the BSC's regular meetings. The BEU shall be provided with a copy of the Official Agenda of each regular or public meeting of the BSC, prior to such meeting. The BEU shall be provided with copies of the official public minutes of each regular BSC meeting as soon as possible after the approval of the said minutes by the BSC.
 - The BEU shall have the right to address the teachers at the first general faculty meeting of the school system.
- **10.3 Information on New Hires**. The BEU shall receive from the Superintendent the names of all new teachers by the opening of school. The Superintendent or his/her designee shall provide the President of the BEU monthly with the name of employees in the

bargaining unit hired during that month, where the new employee has been assigned, the new employee's years of experience and what the new employee's pay scale is, and effective September 2012, including step placement.

- **10.4 Athletic facilities**. Teachers shall have the right to use the athletic facilities of the schools of the Town during the regularly scheduled hours of these facilities.
- **10.5 BEU Days**. Teachers and building representatives of the BEU may request and they may be granted educational leave to attend meetings and workshops held by the local, state, or national associations. The BEU shall be allowed an aggregate of thirty (30) such days for all persons in Units A and B and the Paraprofessional Unit each school year. Up to ten (10) unused days from the previous year will be added to the thirty (30). At no time shall the total days exceed forty (40).

10.6 Agency Service Fee

Effective September 1, 1993, any teacher employed by the BSC shall be subject to an agency service fee unless said teacher is or becomes a members of the BEU. The fee shall equal the cost of collective bargaining and contract administration except in no case shall the fee exceed the dues amount for the current membership year.

Collection of agency service fees, including the collection of delinquent fees shall be solely the responsibility of the BEU, and the BSC shall not be obligated to take any action in regard to the employment of such delinquent employees. Teachers who fail to pay the fee shall not be subject to dismissal or suspension, but the BEU may pursue payment through whatever legal means it deems appropriate.

The BEU agrees to set the amount of the agency service fee and to administer procedures relating to the fees in accordance with all applicable laws and regulations.

Effective with the first pay period in September 2010, the BSC agrees to deduct the agency service fee from the paychecks of employees who authorize such deductions and to forward such fees to the recipient designated by the BEU. (The parties agree to use the attached voluntary authorization form.) The BSC will provide the new employees with the agreed upon authorization form. Completed forms will be available to the BEU in the payroll office.

10.7 Meetings with the Superintendent

Effective September, 2011, the Superintendent and/or his/her designee(s) and the BEU President and/or her/his designee(s) will endeavor to meet once quarterly at a mutually convenient time to discuss issues that have already been addressed with the appropriate supervisor(s) and which in most circumstances have system-wide implications. This section 10.7 shall not be subject to the grievance and arbitration provision of this Agreement.

Article 11: Reductions In Force ("RIF")

The BSC and the BEU reaffirm their commitment to excellence in education and to length of service in the Public Schools of Brookline. In the event that it becomes necessary to reduce the number of employees within the school system, the following shall apply.

- 11.1 Seniority shall be defined as length of continuous service in the Public Schools of Brookline in years, months and days from the most recent date of hire.
 - A. Paid leaves of absence shall count toward the accrual of seniority.
 - B. Approved unpaid leaves of absence of fewer than forty-one (41) consecutive working days shall be counted toward the accrual of seniority.
 - C. Approved unpaid leaves of absence of longer than forty (40) days shall not constitute a break in service, but time spent on such leave shall not count toward accrual of seniority.
 - D. Those employees hired prior to September 1, 1999 will accrue seniority as follows:
 - i. The years in which a person was on leave of absence but worked .2 or more for the entire year will be counted as a year of seniority.
 - ii. Full-time unpaid leaves of absence taken for part of the year will be deducted from seniority and longevity.
 - E. Those employees hired on or after September 1, 1999 will accrue seniority as follows:
 - i. Anyone working .5 or more in a given year will receive one year toward seniority. Anyone working less than .5 but .2 or more for the entire year shall receive .5 for the year they are working toward seniority. This means the difference between their date of hire and longevity date in each of these years would be six months.
 - ii. This arrangement includes people who are working less than full time due to a LOA and those with less than full time PTS.
 - iii. Full-time unpaid leaves of absence taken for part of the year will be deducted from seniority.
- 11.2 For purposes of this Agreement, seniority shall be defined as length of continuous service in the Public Schools of Brookline in years, months and days from the most recent date of hire.
 - A. Paid leaves of absence shall count toward the accrual of seniority.
 - B. Approved unpaid leaves of absence of fewer than forty-one (41) consecutive working days shall be counted toward the accrual of seniority.

- C. Approved unpaid leaves of absence of longer than forty (40) days shall not constitute a break in service, but time spent on such leave shall not count toward accrual of seniority.
- 11.3 In the event it becomes necessary to reduce the number of employees within the school system, then no teacher with professional status shall be laid off if there is a teacher with pre-professional status serving in a position that a teacher with professional status is qualified to fill. In addition, no teacher with more than three (3) full consecutive school years of service in the Public Schools of Brookline as a teacher shall be laid off if there is a teacher with pre-professional status serving in a position where such teacher (with more than three (3) full consecutive school years of service in the Public Schools of Brookline) is either qualified and teaching in the discipline or is qualified and certified/licensed in the discipline.
- 11.4 If, in the application of this Article, it is necessary to determine the seniority order of two or more employees with the same seniority credits, the following procedure shall apply in the following order: (a) a full-time employee shall be deemed senior to a part-time employee; (b) the employee with the most experience as a substitute employee in the discipline will be deemed senior to the other employees; or if necessary, (c) a lottery shall be conducted in the presence of the Assistant Superintendent for Human Resources and the President of the BEU, or their designees, to determine the order of seniority among such employees.
- 11.5 In the event that layoffs of teachers with professional status become necessary, the following procedures shall be employed:
 - A. The least senior teacher in an affected discipline area shall be reduced and subsequent reductions within the affected discipline area shall be in reverse order of seniority.
 - B. If an employee's evaluation, prepared in accordance with Sections 6.2 and 6.3 of the Collective Bargaining Agreement shows that the employee's performance was unsatisfactory during the year preceding the year in which the decision to reduce forces takes place, such employee shall be evaluated again prior to the time that the reduction is implemented. If the employee's performance evaluation continues to be unsatisfactory, then that employee shall be deemed to be the least senior employee in his/her discipline, for purposes of this provision. Upon return to satisfactory performance, the affected employee shall be restored to his/her regular place on the seniority list.
 - C. An employee laid off from an affected discipline area may replace a less senior teacher in another discipline area provided that he/she is certified/licensed and qualified to teach in the new discipline. The following shall be the criteria employed by the Superintendent or his/her designee in determining qualifications:
 - i. prior teaching experience in the new discipline;
 - ii. major and minor fields of college study (as shown by official transcript);
 - iii. recent college-level course work in the new discipline's subject area (s);
 - iv. related experience outside the field of education.

- D. If there is a dispute concerning whether the Superintendent or his/her designee employed the above criteria in the determination of qualifications, then a joint committee, consisting of two (2) BSC members or their designees and two (2) BEU members (appointed by the BEU President) shall review the decision and shall attempt to resolve the dispute. In the event that the joint committee cannot reach agreement, the matter will be referred to expedited arbitration pursuant to Article 11.14 of this provision.
- E. For purposes of this provision, discipline areas shall be as follows:
 - i. Early Education
 - ii. Elementary K-6
 - iii. Performing Arts K-12
 - a. Vocal Music
 - b. Instrumental Music
 - c. Dance
 - d. Drama
 - iv. Specialists K-6
 - a. Literacy
 - b. Mathematics
 - v. Physical Education K-12
 - vi. Computers 7-12
 - vii. Special Education K-12
 - viii. English Language Learners/Transition Bilingual Education/ESL K-12
 - ix. English 7-12
 - x. Mathematics 7-12
 - xi. Social Studies/Political Science 7-12
 - xii. Foreign Language 7-12
 - a. French
 - b. Spanish
 - c. Latin
 - d. German
 - e. Chinese
 - f. Japanese
 - g. Russian

xiii. Science 7-12

- a. Biology
- b. Chemistry
- c. Physics
- d. Earth Science/Physical Science
- e. General Science
- xiv. Home Economics 4-12
- xv. Industrial Arts K-12
- xvi. Visual Arts K-12
- xvii. Occupational Education
 - a. Business education/marketing
 - b. Early childhood education
 - c. Food service careers
 - d. Horticultural/Conservation
- xviii. Guidance K-12
 - xix. School Adjustment Counselors/Social Workers K-12
 - xx. Speech and Language Therapists K-12
 - xxi. School Psychologists K-12
- xxii. Library Science K-12
- xxiii. Health
- xxiv. Nurses
 - a. School Clinic Nurses
 - b. Support Nurses
- xxv. Associate Deans
- xxvi. Occupational Therapists K-12
- xxvii. Physical Therapists K-12
- xxviii. African-American Scholars Leader
 - xxix. Vision Specialists
 - xxx. Lesley Internship Coordinator
 - xxxi. Hearing Specialists (when applicable)
- xxxii. Library Assistants
- xxxiii. Lab Specialists
 - a. Foreign Language Lab
 - b. Humanities Lab
 - c. Math and Science Lab

- 11.6 Whenever an administrator who is a bargaining unit member and who has professional teacher status in the system is reduced, s/he shall be credited with all seniority earned as a teacher and allowed to replace the least senior teacher in the discipline area of most recent employment providing that the administrator has greater seniority than the teacher so affected. If no teacher in the specific discipline area is less senior than the administrator, then the administrator may exercise bumping rights in accordance with Article 11.3 of this provision.
- 11.7 It is understood that if it can be demonstrated that a particular employee within a discipline is the only person in the discipline qualified to teach a course or provide a specialized service of a professional/educational nature, that employee may be retained regardless of seniority.

In determining whether such an employee is qualified the following shall be the criteria employed by the BSC:

- a. prior teaching experience in the new area/level (K-6/7-12)
- b. major and minor fields of college study (as shown by official transcript);
- c. recent college-level course work in the new area/level's subject area (s);
- d. related experience outside the field of education.

If a teacher in a K-12 discipline area, a guidance counselor, school adjustment counselor, school psychologist or speech and language therapist is moving from the 7-12 area/level to the K-6 area/level or vice versa as the result of a RIF, the Superintendent shall have the right to have an evaluation made in the first year the employee assumes the new position, that evaluation being based upon a set of goals and objectives agreed upon between the BEU and the Superintendent prior to the person assuming the position. This evaluation procedure shall be based on the same progressive procedures currently used to judge the status of teachers with pre-professional status. The Superintendent and BEU will arrange in advance to provide any counseling/assistance which the parties deem helpful.

If the evaluation of such employee at the end of the evaluation year is unsatisfactory, the Superintendent may, at his option, initiate termination or RIF procedures or may extend the evaluation period to a second year, with the option at the end of that time to initiate termination procedures. The employee may elect to be laid off prior to any termination procedure initiated by the Superintendent. It is further understood that this provision does not apply to any employee who works across levels and is then assigned to a single area/level as defined above.

- 11.8 Reductions in individual full-time equivalencies will be accomplished by seniority, consistent with the Sections set forth above.
- 11.9 No later than October 30 of each school year, the Superintendent or his/her designee shall provide each member of the bargaining unit with an up-to-date seniority list by discipline. Any challenges to the accuracy of the list must be registered, in writing, to the Assistant Superintendent for Human Resources before December 1.

The BSC agrees that any published seniority list will also contain the names of all preprofessional status staff members.

- 11.10 When a RIF is to take place, the Superintendent or his/her designee shall give written notice to the affected teachers with professional status by certified mail, return receipt requested, no later than May 15 preceding the effective date of the reduction. If, however, the RIF of any individual teacher are rescinded before the effective date of the reduction and the teacher is to be reemployed, the Assistant Superintendent for Human Resources shall so inform the affected teacher in writing.
- 11.11 An employee affected by this Article will, for 27 months for employees with more than 5 years of service, and 15 months for employees with 5 years of service or less following the effective date of layoff, retain recall rights to fill a vacancy within the employee's discipline. In the event of a vacancy, the most recently affected employee who has recall rights within the discipline will be offered the position created by the vacancy up to the level (part-time or full-time) such employee held at the time of layoff. Recall notices shall be sent by certified or registered mail to the employees.

It shall be the responsibility of the laid-off employee to keep the Assistant Superintendent for Human Resources informed of his/her current address and telephone number.

11.12 An employee must notify the Superintendent within fourteen (14) calendar days of the actual receipt of recall notice of his/her intent to accept or reject recall. Except as set forth in the following paragraph, if an employee rejects recall or fails to notify the Superintendent within the specified period, the employee shall forfeit all rights and benefits provided for in this Article. Unless otherwise mutually agreed in writing, an employee who accepts recall must begin work on the date set forth in the recall notice or within 14 calendar days, whichever is later. An employee who accepts recall shall retain all service credits and benefits accrued through the effective date of the layoff. While on recall status, laid off employees shall be given preference on the substitute teacher list.

If an employee who is recalled is unable to return within the time set forth above because the employee holds a job in another school system and is committed to complete the school year there, or because the employee is enrolled as a full-time student in a college or university, such employee shall remain extended by the period of time between the date when the employee received the notice of recall and the date when that employee's school year ends. If such employee is subsequently recalled and accepts recall pursuant to the preceding paragraph, s/he shall retain seniority accrued as of the date of layoff and shall accumulate seniority for any period of the layoff in excess of fifteen (15) months.

11.13 To the extent permitted by applicable law, laid-off employees may continue group health and life insurance coverage during the recall period, as provided by the BSC to members of the bargaining unit, by reimbursing the BSC for the premium costs. Failure to forward premium payments to the BSC or refusal to return to employment upon recall will terminate this option.

11.14 Any disputes over the meaning, interpretation or application of this article shall be resolved in accordance with the Grievance and Arbitration provisions of Article 9; provided, that in the event of a factual dispute over the seniority standing of two or more employees or under 11.4 above, such dispute shall be resolved in accordance with the expedited arbitration procedures of the American Arbitration Association.

Article 12: Sexual Harassment Policy

The parties agree that the Sexual Harassment Policy and Procedures adopted by the BSC on June 19, 2003 is hereby incorporated into this Agreement and as such is enforceable under Article 9.

Article 13: Duration

This contract shall take effect as of September 1, 2011 and shall continue and remain in full force and effect through August 31, 2014.

The foregoing Contract was approved by the BSC on November 3, 2011 and ratified by the BEU on October, 2011 and is identified as such Contract by the signers as set forth below.

The parties intend that negotiations for a successor contract shall commence no later than November 1, 2013.

For the BEU	For the BSC
	-
	-
	-
	_
	-
	-
	-
	-
	_
	-
	_
	_

SIDE LETTERS OF AGREEMENT

The BSC and the BEU enter into the following agreements. Other side letters have been negotiated in previous years. The fact that they are not included does not invalidate them.

- 1. New Miscellaneous. The BSC and the BEU recognize that from time to time miscellaneous day-to-day issues arise. Some examples are the need for use of a private room, telephone, computer, etc. or the fact that parking spaces reserved for BPS employees are unlawfully being used by outsiders. The BSC and the BEU agree that if such an issue arises, the teacher should go to the appropriate representative of the BSC, e.g. a school principal. In the event the matter is not resolved to the satisfaction of the teacher, he/she may refer it to a joint committee consisting of one representative each designated by the BSC and BEU. It is also understood that this is not intended to add or subtract from the rights and obligations of the parties under the collective bargaining agreement.
- 2. **Definition of Domestic Partners.** The definition of domestic partners shall be as provided by the prior Town Meeting vote, any subsequent Town Meeting vote or any subsequently applicable state law, and with the further understanding that the cost of premiums will be shared by the Town and the employee as otherwise may be applicable to individual or family coverage. Any adverse tax burden as may be imposed under federal or state law relating to the taxability of domestic partner health insurance coverage will be borne by the individual employee.
- 3. Time and Learning Implementation. A Labor-Management Committee, made up of an equal number of representatives of the BEU and BSC will be convened to address specific concerns that may arise due to the implementation of the Time and Learning agreement. This committee will meet on a periodic basis to review these concerns: core hours of the student school day, structured learning time, after-school meetings, implementation by the principals of the Time and Learning Agreement; extended library hours; effect on school nurses; equitable planning time among elementary schools and grade levels; and other issues that the representatives may raise from time to time.

4. Relicensure

This letter between the parties outlines the principles by which the parties here agree to address relicensure issues flowing from the BSC's role, assigned by the DESE, in managing the process of approving or disapproving IPDP's and the consequences to employees which flow from disapproval. The terms of this letter are not subject to the grievance and arbitration procedure or statutory arbitration or litigation.

RELICENSURE POLICY

The following relicensure policy to be issued by the Superintendent will govern implementation of the Relicensure Regulations adopted by DOE and shall not be subject to the grievance and arbitration procedures. Pursuant to the requirements of Recertification Regulations, CMR 44.00, adopted by the Massachusetts Board of Education on October 26, 1999, the parties agree as follows:

- A. The Principal or Vice-Principal shall be responsible for approving the Individual Professional Development Plans (IPDPs) of educators at the elementary level. The Headmaster, Deans, Assistant Headmaster, Program Coordinator or Curriculum Coordinator, at the discretion of the Headmaster, shall be responsible for approving the IPDPs of educators at the secondary level. In the case of shared staff, the Superintendent shall designate the supervisor who is responsible for the approval of the IPDP for the duration of the educator's relicensure cycle.
- B. The IPDP will be submitted on the form found in Appendix D.
- C. The IPDPs which are not inconsistent with school or district goals as outlined in Superintendent's annual goals and school annual goals shall be approved.
- D. The supervisor responsible for approval will return the approved IPDP to the individual educator no later than 25 work days following receipt of the IPDP. This timeline also applies to disapproval and final endorsement. IPDPs not returned in 25 days will be considered approved.
- E. In the case of disapproval, the supervisor will state specific reasons for disapproval in writing to the educator and will specify what changes need to be made in the IPDP to gain approval. This provision will also apply in the case of any amendment to the plan at the time of the two-year review during the relicensure cycle.
- F. An educator on leave of absence shall be granted three (3) months from the date of return to work to obtain approval of the IPDP.
- G. Educators who do not intend to recertify are not required to seek approval of an IPDP.
- H. In the event that educators are no longer required by regulation to obtain initial approval or final endorsement of an IPDP for relicensure, this Memorandum shall be declared null and void.
- I. These procedures, the attached form and the school and district goals will be sent to all teachers and administrators in a memo from the Superintendent by September 20 of each year.
- J. Any educator whose IPDP has been disapproved shall have the right to a meeting with the Superintendent and at such meeting shall have the right to be accompanied by a BEU representative who may be an MTA staff representative. Involvement of attorneys is not contemplated by this provision.

5. CORI Agreement

The "CORI Law" (M.G.L. c. 71, Sec. 38R) requires the BSC to obtain Criminal Offender Record Information ("CORI") on all current and prospective employees who may have direct and unmonitored contact with children. The BSC and the BEU agree as follows:

- A. The Superintendent or his/her designee will limit his/her request for CORI's to once every three years for any individual employee except if there is a reasonable suspicion that an employee has been arraigned on charges that pose a hazard to children. If such an out of cycle CORI is issued, then the three-year cycle starts anew.
- B. Although it is understood that CORI constitutes personnel record information, CORI's will not be filed in any individual employee's personnel file.
- C. All CORI's will be retained in one secure location as determined by the Assistant Superintendent for Human Resources and only one copy of any individual CORI shall be retained by the Committee.
- D. The BSC will make every attempt to assure that CORI's are handled only by the Assistant Superintendent for Human Resources and/or his/her administrative assistant. Access to any individual CORI will be consistent with the provisions of Chapter 385 of the Acts of 2002. The Assistant Superintendent for Human Resources will establish a log in order to record the name and title of anyone accessing records and the records accessed.
- E. Upon receiving a written request from an employee for a copy of his or her CORI on file with the Public Schools of Brookline, the Assistant Superintendent for Human Resources or his/ her administrative assistant shall provide the employee with a copy of such CORI.
- F. It is understood that an employee has the right to challenge a CORI pursuant to the statute and to M.G.L. Ch. 149 Section 52C. It is further understood that the Committee and/or its designees will comply with all contractual requirements that employees be notified within twenty school days regarding any derogatory material that comes to the attention of the Committee and/or its designees.
- G. Unless treated as a derogatory material, no CORI will be retained more than three years.
- H. The BSC shall access sealed record information only as permitted by law.
- I. The CORI Policy was adopted by the BSC on June 19, 2003. It is understood by both parties that this Policy and the Procedures embodied in this Agreement are the only CORI Policy and Procedures that apply to employees represented by the BEU.

6. K-8 Teachers (effective September 1, 2007)

The following shall apply to those teachers subject to provisions of Section 4.3.A.ii: "In the first 15 minutes of the school day, teachers will provide instruction which may include structured learning activities. Teachers will be in their classrooms sufficiently prior to the start of the student day to welcome students and provide students with access to the classroom so that instruction will begin at the start of the student day."

APPENDIX A: - Salary Schedules

A.1 Teachers' Salary Schedules

Effective at 11:59 p.m. on August 31, 2011

(\$500 increase)

STEP	BACHELOR'S DEGREE	MASTER'S DEGREE	MASTER'S DEGREE + 30	MASTER'S DEGREE + 45	DOCTOR'S DEGREE
1	45,593	48,545	51,600	53,119	54,570
2	48,039	51,065	54,196	55,714	57,264
3	50,484	53,588	56,795	58,313	59,958
4	52,922	56,109	59,390	60,910	62,649
5	55,368	58,630	61,984	63,503	65,339
6	57,813	61,154	64,580	66,103	68,035
7	60,255	63,676	67,177	68,696	70,725
8	62,701	66,196	69,779	71,296	73,421
9	65,145	68,717	72,374	73,895	76,114
10	67,593	71,241	74,973	76,489	78,808
11	70,031	73,765	77,569	79,088	81,500
12	72,479	76,283	80,166	81,688	84,194
13	74,814	78,808	82,760	84,281	86,885
14		81,919	85,359	86,878	89,581
15			88,470	89,991	92,271
16					94,610

Effective on September 1, 2012

(2% increase)

STEP	BACHELOR'S DEGREE	MASTER'S DEGREE	MASTER'S DEGREE + 30	MASTER'S DEGREE + 45	DOCTOR'S DEGREE
1	\$46,505	\$49,516	\$52,632	\$54,181	\$55,661
2	\$49,000	\$52,086	\$55,280	\$56,828	\$58,409
3	\$51,494	\$54,660	\$57,931	\$59,479	\$61,157
4	\$53,980	\$57,231	\$60,578	\$62,128	\$63,902
5	\$56,475	\$59,803	\$63,224	\$64,773	\$66,646
6	\$58,969	\$62,377	\$65,872	\$67,425	\$69,396
7	\$61,460	\$64,950	\$68,521	\$70,070	\$72,140
8	\$63,955	\$67,520	\$71,175	\$72,722	\$74,889
9	\$66,448	\$70,091	\$73,821	\$75,373	\$77,636
10	\$68,945	\$72,666	\$76,472	\$78,019	\$80,384
11	\$71,432	\$75,240	\$79,120	\$80,670	\$83,130
12	\$73,929	\$77,809	\$81,769	\$83,322	\$85,878
13	\$76,310	\$80,384	\$84,415	\$85,967	\$88,623
14		\$83,557	\$87,066	\$88,616	\$91,373
15			\$90,239	\$91,791	\$94,116
16					\$96,502

- All steps beyond Step 5 on the Bachelor's lane shall be eliminated for employees who commenced employment in Unit A on or after Sept. 1, 2012 with PSB.
- Employees who commenced employment in Unit A in or before the 2011-2012 school year and who were on Step 5 or below on the Bachelor's lane in the 2011-2012 school year shall not progress past Step 5 in the Bachelor's column.
- Employees who commenced employment in Unit A in or before the 2011-2012 school year and who were on Step 6 or above during the 2011-2012 school year may continue through the steps in effect in the Bachelor's column.
- Nurses, automotive teachers, construction teachers and food services teachers who are not required to have a degree above the Bachelor's level may progress through all steps in the Bachelor's lane.
- Effective Sept. 1, 2013 add a new Bachelor's Step 14 at the rate of \$2,500 above Step 13.
- Effective at 11:59 p.m. on Aug. 31, 2014 add a new Bachelor's Step 15 at the rate of \$2,500 above Step 14.

Effective at 12:01 a.m. on September 1, 2013

(2% increase)

STEP	BACHELOR'S DEGREE	MASTER'S DEGREE	MASTER'S DEGREE + 30	MASTER'S DEGREE + 45	DOCTOR'S DEGREE
1	\$47,435	\$50,506	\$53,685	\$55,265	\$56,775
2	\$49,980	\$53,128	\$56,386	\$57,965	\$59,577
3	\$52,524	\$55,753	\$59,090	\$60,669	\$62,380
4	\$55,060	\$58,376	\$61,789	\$63,371	\$65,180
5	\$57,605	\$60,999	\$64,488	\$66,069	\$67,979
6	\$60,149	\$63,625	\$67,189	\$68,774	\$70,784
7	\$62,689	\$66,249	\$69,891	\$71,471	\$73,582
8	\$65,234	\$68,870	\$72,598	\$74,176	\$76,387
9	\$67,777	\$71,493	\$75,298	\$76,880	\$79,189
10	\$70,324	\$74,119	\$78,002	\$79,579	\$81,992
11	\$72,860	\$76,745	\$80,703	\$82,283	\$84,793
12	\$75,407	\$79,365	\$83,405	\$84,988	\$87,595
13	\$77,836	\$81,992	\$86,104	\$87,686	\$90,395
14		\$85,229	\$88,808	\$90,388	\$93,200
15			\$92,044	\$93,627	\$95,999
16					\$98,432

Effective at 12:05 a.m. on September 1, 2013

STEP	BACHELOR'S DEGREE	MASTER'S DEGREE	MASTER'S DEGREE + 30	MASTER'S DEGREE + 45	DOCTOR'S DEGREE
4		\$50,506	\$53,685	\$55,265	\$56,775
1	\$47,435	\$53,128	\$56,386	\$57,965	\$59,577
2	\$49,980	\$55,753	\$59,090	\$60,669	\$62,380
3	\$52,524	\$58,376	\$61,789	\$63,371	\$65,180
4	\$55,060	\$60,999	\$64,488	\$66,069	\$67,979
5	\$57,605	\$63,625	\$67,189	\$68,774	\$70,784
6	\$60,149	\$66,249	\$69,891	\$71,471	\$73,582
7	\$62,689	\$68,870	\$72,598	\$74,176	\$76,387
8	\$65,234	\$71,493	\$75,298	\$76,880	\$79,189
9	\$67,777	\$74,119	\$78,002	\$79,579	\$81,992
10	\$70,324	\$76,745	\$80,703	\$82,283	\$84,793
11	\$72,860	\$79,365	\$83,405	\$84,988	\$87,595
12	\$75,407	\$81,992	\$86,104	\$87,686	\$90,395
13	\$77,836	\$85,229	\$88,808	\$90,388	\$93,200
14	\$80,336	\$87,729	\$92,044	\$93,627	\$95,999
15			\$94,544	\$96,127	\$98,432
16					\$100,932

- Effective 12:05 a.m. on Sept. 1, 2013, add a new top step at \$2.500 above the prior top step in each of the following columns: Master's, Master's +30, Master's +45, and Doctorate Degree.
- Effective at 11:59 p.m. on August 31, 2014, add a new top step at \$2,500 above the prior top step in each of the following columns: Master's, Master's +30, Master's +45, and Doctorate Degree.
- Effective Sept. 1, 2013, eliminate Step 1 from each of the following columns: Master's, Master's +30, Master's +45, and Doctorate Degree.
- <u>Double Step:</u> Effective Sept. 1, 2013, employees who have a Master's Degree or above who have completed seven (7) consecutive work years of service with the required license in Unit A in the PSB and who are beginning the eighth consecutive work year of licensed service shall take a double step. (The Double Step shall not apply to any employee who is beyond his/her 8th consecutive year of service as of Sept. 1, 2013. Unpaid leaves of absence pursuant to Section 5.6 of Leaves of Absence without Pay represent a break of continuous service for the purposes of this Double Step. A leave of absence of more than one year pursuant to 5.9 Child Rearing Leave shall constitute a break in service for the purposes of this Double Step. A leave of absence of up to one year pursuant to Section 5.9 Child Rearing Leave shall not constitute a break in service for the purpose of this Double Step but said leave shall not count towards satisfying the seven (7) consecutive work year requirement.

Examples:

- A. Teacher A has a Master's Degree. She has served as a teacher in the Public Schools of Brookline for seven consecutive years with the requisite license for her position. She was at Step 7 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is commencing her 8th consecutive year of service with the requisite license. She takes a double step and is placed on Step 9.
- B. Teacher B has a Master's Degree. She has served as a teacher in the Public Schools of Brookline for ten years with the requisite license for her position. She was at Step 10 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is commencing her 11th consecutive year of service. She is placed on Step 11.
- C. Teacher C has a Master's Degree. She served as a teacher in the Public Schools of Brookline for six years then took a leave of absence (other than for Child Rearing pursuant to 5.9) for one year without pay. She then returned to work and has worked for one year (2012-2013 academic year). At all times she had the requisite license for her position. She was at Step 7 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is on Step 8.
- D. Teacher D has a Master's Degree. She has served as a teacher in the Public Schools of Brookline for seven consecutive years. She worked under a waiver for her first year in the District and has had the requisite license for her position starting with her second year. She was at Step 7 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is commencing her 8th consecutive year of service. She is placed on Step 8. On September 1, 2014, she is commencing her 9th consecutive year of service and placed on Step 10 because she has been licensed for 7 consecutive years.
- E. Teacher E has a Master's Degree. She served as a teacher in the Public Schools of Brookline for six years then took a leave of absence for <u>one</u> year without pay for Child Rearing pursuant to 5.9. She then returned to work and has worked for one year (2012-2013 academic year). At all times she had the requisite license for her position. She was at Step 7 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is on Step 9.
- F. Teacher F has a Master's Degree. She served as a teacher in the Public Schools of Brookline for six years then took a leave of absence for two years without pay for Child Rearing pursuant to 5.9. She then returned to work and has worked for one year (2012-2013 academic year). At all times she had the requisite license for her position. She was at Step 7 in the Master's Degree column in the 2012-2013 work year. On September 1, 2013 she is on Step 8.

Effective at 11:59 p.m. on Aug 31, 2014

STEP	BACHELOR'S DEGREE	MASTER'S DEGREE	MASTER'S DEGREE + 30	MASTER'S DEGREE + 45	DOCTOR'S DEGREE
	DEGREE	DEGREE	DEGREE + 30	DEGREE + 45	DEGREE
1	\$47,435	\$53,128	\$56,386	\$57,965	\$59,577
2	\$49,980	\$55,753	\$59,090	\$60,669	\$62,380
3	\$52,524	\$58,376	\$61,789	\$63,371	\$65,180
4	\$55,060	\$60,999	\$64,488	\$66,069	\$67,979
5	\$57,605	\$63,625	\$67,189	\$68,774	\$70,784
6	\$60,149	\$66,249	\$69,891	\$71,471	\$73,582
7	\$62,689	\$68,870	\$72,598	\$74,176	\$76,387
8	\$65,234	\$71,493	\$75,298	\$76,880	\$79,189
9	\$67,777	\$74,119	\$78,002	\$79,579	\$81,992
10	\$70,324	\$76,745	\$80,703	\$82,283	\$84,793
11	\$72,860	\$79,365	\$83,405	\$84,988	\$87,595
12	\$75,407	\$81,992	\$86,104	\$87,686	\$90,395
13	\$77,836	\$85,229	\$88,808	\$90,388	\$93,200
14	\$80,336	\$87,729	\$92,044	\$93,627	\$95,999
15	\$82,836	\$90,229	\$94,544	\$96,127	\$98,432
16			\$97,044	\$98,627	\$100,932
17					\$103,432

A.2 Library Assistants' Salary Schedules

Effective at 11:59 p.m. on August 31, 2011 (\$500 increase)

STEP	NO DEGREE	BACHELOR'S DEGREE	MASTER'S DEGREE
JILF	DLUNLL	DEGINEE	DEGINEE
1	28,003	29,265	31,119
2	29,693	30,951	32,706
3	31,364	32,631	34,299
4	33,055	34,316	35,886
5	34,737	35,994	37,487
6	35,765	37,025	38,515
7	36,807	38,105	39,640
8	38,418	39,775	41,379

Effective on September 1, 2012

(2% increase)

	NO	BACHELOR'S	MASTER'S
STEP	DEGREE	DEGREE	DEGREE
1	\$28,563	\$29,850	\$31,741
2	\$30,287	\$31,570	\$33,360
3	\$31,991	\$33,284	\$34,985
4	\$33,716	\$35,002	\$36,604
5	\$35,432	\$36,714	\$38,237
6	\$36,480	\$37,766	\$39,285
7	\$37,543	\$38,867	\$40,433
8	\$39,186	\$40,571	\$42,207

Effective at 12:01 a.m. on September 1, 2013

(2% increase)

	NO	BACHELOR'S	MASTER'S
STEP	DEGREE	DEGREE	DEGREE
1	\$29,134	\$30,447	\$32,376
2	\$30,893	\$32,201	\$34,027
3	\$32,631	\$33,949	\$35,685
4	\$34,390	\$35,702	\$37,336
5	\$36,140	\$37,448	\$39,001
6	\$37,210	\$38,521	\$40,071
7	\$38,294	\$39,644	\$41,241
8	\$39,970	\$41,382	\$43,051

A.3 Lab Specialists' Salary Schedules

Effective at 11:59 p.m. on August 31, 2011

(\$500 increase)

	· · · · · · · · · · · · · · · · · · ·	,
	BACHELOR'S	MASTER'S
STEP	DEGREE	DEGREE
1	35,354	37,644
2	37,543	39,932
3	39,933	42,225
4	42,225	44,514
5	44,514	46,806
6	46,805	49,093
7	49,093	51,382
8	50,537	52,894
9	52,766	55,229

Effective on September 1, 2012

(2% increase)

	BACHELOR'S	MASTER'S
STEP	DEGREE	DEGREE
1	36,061	38,397
2	38,294	40,731
3	40,732	43,070
4	43,070	45,404
5	45,404	47,742
6	47,741	50,075
7	50,075	52,410
8	51,548	53,952
9	53,821	56,334

Effective at 12:01 a.m. on September 1, 2013

(2% increase)

	BACHELOR'S	MASTER'S
STEP	DEGREE	DEGREE
1	36,782	39,165
2	39,060	41,545
3	41,546	43,931
4	43,931	46,312
5	46,312	48,697
6	48,696	51,076
7	51,076	53,458
8	52,579	55,031
9	54,898	57,460

APPENDIX B: Extra Compensation

- **B.1** Compensatory Time. Any employee covered by this Agreement who is required to work longer than the days stipulated in Article 4.1 shall have the option of being paid the per diem rate of his/her annual salary for each additional day or of receiving compensatory time off at a date mutually agreed upon by the employee and his/her supervisor.
- **B.2** Extra Compensation for Specific Positions. The following personnel shall receive a per annum stipend in addition to their regular salaries as follows:

<u>Position</u>	Effective 8/31/11	Effective 9/1/12	Effective 9/1/13
OT Coordinator	\$4,085	\$4,167	\$4,250
School Adjustment Counsel	or \$ 448	\$ 457	\$ 466
Associate Deans	\$3,060	\$3,000	\$3,030

The following personnel shall work additional days than stipulated in Article 4.1 and shall be paid at the per diem rate of his/her annual salary:

<u>Position</u>	Number of Days
Time and Learning Coordinator	7
International Student Adv.	6
Associate Dean	7
African American Scholars Lead Teacher	7

B.3 AP Science Teachers. Teachers who are assigned to extra lab period(s) in AP science, in addition to their normal working assignments, shall receive the following stipends:

Effective Date	<u>1 extra lab</u>	<u>2 extra labs</u>
September 1, 2007	\$ 700	\$ 1,400
September 1, 2012	\$ 714	\$ 1,428
September 1, 2013	\$ 728	\$ 1,457

It is understood that to the extent such teachers have been released from contractual duties in the past, they will now be required to perform all duties pursuant to the contract.

Effective between September 1, 2010 and August 31, 2011, the parties agree that upon the written request of either party to reopen on the issue of compensation in Appendix B.3 for AP Science teachers.

B.4 After School Sports Program. Teachers participating in the elementary intramural sports program will receive the following amounts per session (one hour):

Effective Date	<u>Amount</u>
August 31, 2011 at 11:59 p.m.	\$34.13
September 1, 2012	\$34.81
September 1, 2013	\$35.51

Teachers participating in the elementary extramural sports program will receive the following amounts per session (one hour):

Effective Date	<u>Amount</u>
August 31, 2011 at 11:59 p.m.	\$37.53
September 1, 2012	\$38.28
September 1, 2013	\$39.05

Teachers participating in the high school intramural sports program will receive the following amounts per session (90 minutes):

Effective Date	<u>Amount</u>
August 31, 2011 at 11:59 p.m.	\$51.20
September 1, 2012	\$52.22
September 1, 2013	\$53.27

- **B.5 Mileage Allowance**. Teachers who are required to use their automobiles for authorized employment-related travel, shall be compensated at the per mileage allowance rate of the Town of Brookline.
- **B.6** Additional Funds for Coaching. In allocating any additional funds, a Study Committee comprised of representatives of the BEU and the BSC shall meet to address certain inequities, make coaching stipends more competitive with other comparable communities and add new positions. It is understood that the BSC is not obligated to allocate any funds beyond those set forth in the table of rates entitled "Appendix B-8".
- **B.7 Other Extra Compensation**. Future positions which do not involve direct supervision or advice to students of the type described in Appendix B.8, will be included in App. B.8 if the entire compensation for the position comes directly from the BSC budget; if the position is permanent, i.e., the BSC determines that it will continue in existence for more than two years and if the compensation for such position is established at \$500 per year or more. Openings in such positions shall be posted. App. B.8 positions whose stipends are less than \$500 will be listed when they become permanent.

If the stipend for a position is \$1,000 or more <u>per annum</u>, the BSC will negotiate the stipend if and when it becomes an App. B-8 position. In establishing a stipend for a position, the BSC will consider, insofar as is practicable, the expected length of time and the hourly rates set forth in the next paragraph. The BSC, however, shall not be bound to establish the stipend in each case based on these factors. It is understood that this is not applicable to any stipends not funded in full by the BSC.

Any employee covered by this Agreement who is requested to work longer than the number of days stipulated as the work year in Article 4.1 to perform tasks such as staff development, curriculum development and in-service workshops shall be paid at the following hourly rates (excluding non-working lunch).

Effective 8/31/11

	<u>at 11:59 p.m.</u>	Effective 9/1/12	Effective 9/1/13
For Participants	\$30.60	\$31.21	\$31.84
For Leaders	\$35.70	\$36.41	\$37.14

With respect to positions other than those which are B.8 positions and those tasks which are described in the preceding paragraph, existing practices shall remain in effect.

B.8 Coaches and Advisors

A. Coaching Stipends

Effective August 31, 2011 at 11:59 p.m.

(1% increase)

(275					
Class	Step 1	Step 3	Step 5	Step 10	Step 15
1	\$6,849	\$7,412	\$8,057	\$8,863	\$9,668
2	\$4,794	\$5,189	\$5,639	\$6,204	\$6,768
3	\$4,246	\$4,596	\$4,995	\$5,494	\$5,994
4	\$3,425	\$3,707	\$4,028	\$4,432	\$4,834
5	\$2,533	\$2,743	\$2,981	\$3,279	\$3,577
6	\$1,370	\$1,482	\$1,611	\$1,773	\$1,933
7	\$686	\$740	\$805	\$887	\$967

Effective September 1, 2012

(2% increase)

	(270 11101 04000)					
Class	Step 1	Step 3	Step 5	Step 10	Step 15	
1	\$6,986	\$7,560	\$8,218	\$9,040	\$9,861	
2	\$4,890	\$5,293	\$5,752	\$6,328	\$6,903	
3	\$4,331	\$4,688	\$5,095	\$5,604	\$6,114	
4	\$3,494	\$3,781	\$4,109	\$4,521	\$4,931	
5	\$2,584	\$2,798	\$3,041	\$3,345	\$3,649	
6	\$1,397	\$1,512	\$1,643	\$1,808	\$1,972	
7	\$700	\$755	\$821	\$905	\$986	

Effective September 1, 2013

(2% increase)

(=,					
Class	Step 1	Step 3	Step 5	Step 10	Step 15
1	\$7,126	\$7,711	\$8,383	\$9,221	\$10,059
2	\$4,988	\$5,399	\$5,867	\$6,455	\$7,041
3	\$4,418	\$4,782	\$5,197	\$5,716	\$6,236
4	\$3,563	\$3,857	\$4,191	\$4,611	\$5,029
5	\$2,635	\$2,854	\$3,101	\$3,411	\$3,722
6	\$1,425	\$1,542	\$1,676	\$1,845	\$2,011
7	\$714	\$770	\$838	\$923	\$1,006

- Class 1: Head Coach for Football
- Class 2: Head Coaches for Basketball (2), Ice Hockey (2), Indoor Track (2), Outdoor Track (2); Fall Trainer, Winter Trainer
- Class 3: Head Coaches for Baseball, Softball, Soccer (2), Field Hockey, Volleyball (2), Wrestling, Gymnastics, Lacrosse (2), Crew, Swimming (2); Spring Trainer; Assistant Coaches for Football (2-offensive and defensive coordinators)
- Class 4: Head Coaches for Skiing, Golf (2), Tennis (2), Cross-Country (2), Rugby; Assistant Coaches for Football (3 JV, Head and Asst. Freshman)
- Class 5: Head Coach for Sailing; Assistant Coaches for Basketball (Fr-2), Wrestling, Baseball (2), Softball (2), Soccer (4), Gymnastics, Field Hockey (2), Lacrosse (4), Swimming (2), Crew, Basketball (JV-2), Indoor Track (2), Outdoor Track (2), Volleyball (2), Ice Hockey (4); Equipment Manager (3)
- Class 6: Head Coach for Cheerleaders (2); Assistant Coaches for Tennis (2), Golf (B), Volleyball (2), Skiing
- Class 7: Assistant Coach for Cheerleaders (2)

B. Advisor Stipends

Effective August 31, 2011 at 11:59 p.m.

(1% increase)

Class	Step 1	Step 5
1	\$4,245	\$4,596
2	\$3,265	\$3,534
3	\$2,286	\$2,473
4	\$1,568	\$1,697
5	\$1,177	\$1,272
6	\$ 784	\$ 848
7	\$ 554	\$ 610

Effective September 1, 2012

(2% increase)

Class	Step 1	Step 5
1	\$4,330	\$4,688
2	\$3,330	\$3,605
3	\$2,332	\$2,522
4	\$1,599	\$1,731
5	\$1,201	\$1,297
6	\$ 800	\$ 865
7	\$ 565	\$ 622

Effective September 1, 2013

(2% increase)

Class	Step 1	Step 5
1	\$4,416	\$4,782
2	\$3,397	\$3,677
3	\$2,378	\$2,573
4	\$1,631	\$1,766
5	\$1,225	\$1,323
6	\$ 816	\$ 882
7	\$ 576	\$ 635

- Class 1: Murivian, BHS Drama Director, AP Exam Coordinator
- Class 2: Government Liaison, Scholarship Committee, 7/8 Extramural Sports Coordinator, BHS Musical Director, Town-wide 7/8 Band Director, Town-wide 7/8 Jazz Band Director, Town-wide Girls Chorus Director, Town-wide Boys Chorus Director, Town-wide Youth Orchestra Director
- Class 3: Sagamore, Cadence, BHS Band Director, BHS Chorus Director, BHS Musical Music, BHS Play Director, BHS Freshman Play Director, BHS State Drama Festival Director, Town-wide Jazz Band Director, Town-wide 7/8 Concert Band Assistant, Town-wide Girls Chorus Assistant, Town-wide Grade 6 Jazz Band Director, Town-wide Elementary Orchestra Director, Elementary Drama (8)
- Class 4: Black Awareness Club, Refractions, Senior Class, Gay-Straight Alliance, National Honor Society, Math Team, Science Team, Latin Club, Latino American Club, Jazz Band, Asian Students Club, Russian Club, Math League Coordinator, Debating Club, BHS Drama Construction, BHS Manager of Theater
- Class 5: Math League (8), Math Counts (8), Intramural Sports Coordinator (2, BHS and Grades 4-8), Grade 4-8 Building Sports Coordinators (8), Mock Trial, Students Action for Justice and Education, BHS Dance Ensemble Director, BHS Orchestra Director, Elementary Chorus (2)
- Class 6: Ski Club, Badminton Club, Chess Club, Youth Engaged in Service,
 Oceanography Club, Connections, Environmental Action, Film Club,
 Habitat for Humanity, Hip Hop Lady Warriors, Japanese Animation Fan
 Club, Model UN Club, Philosophy Club, Robotics Club, SADD, Junior
 Class, Comedy Workshop, BHS Musical Lights, BHS Scenery Painting,
 Musical Choral Director
- Class 7: Police Academy, Freshman Class, Sophomore Class

C. Other Advisors

<u>Position</u>	Effective 8/31/11		
	at 11:59 p.m.	Effective 9/1/12	Effective 9/1/13
Freshman Homeroom	\$589	\$601	\$613
Teacher Advisors			
Teacher Mentors	\$612	\$624	\$637
Mentor Facilitators	\$323+\$32/pair	\$329+\$32/pai	ir \$336+32/pair
BHS Treasury	\$5,492	\$5,602	\$5,714
Middle School Sports Coordinate	or \$3,351	\$3,602	\$3,674
Professional Development Counc	eil \$510	\$520	\$531

APPENDIX C: Guidelines

The BSC and the BEU recognize the following guidelines as statements of desirable objectives and procedures. The BSC in exercising its power and performing its duties shall always give consideration to any relevant guidelines set forth in this Appendix.

- **C.1** Class Size. Under normal circumstances no class should be composed of more than 25 students.
- **C.2 Teacher Facilities**. The BSC should endeavor to provide:
 - 1. Teaching stations with:
 - adequate lighting, heat, and ventilation
 - a desk and a chair
 - resources supportive of modern teaching and learning procedures e.g. room darkening shades
 - filmstrip, film, and film loop projectors
 - appropriate software
 - 2. Securable storage facilities for:
 - files
 - books, tapes, etc.
 - electronic hardware
 - personal property
 - 3. Professional Resource and Work Areas:
 - a. Elementary Schools:

One work room for every twelve teachers, with room to be equipped with:

- 1 primary typewriter
- 1 manual typewriter
- 1 electric typewriter
- tables, chairs
- duplicating machine, electric
- 1 dry copier per building
- securable storage for shared resources
- preauditing and previewing facilities
- professional resources and periodicals

b. <u>High School</u>

The equipment listed above under a. in:

- either individual work areas for each discipline or
- shared centers for the disciplines or
- combined student-teacher resource areas for each discipline

- 4. Teacher Lounge with:
 - telephone
 - availability of restrooms
 - availability of light refreshment
 - practical, yet comfortable furniture
- 5. Communication system that is efficient enough to reach all, yet selective enough not to be unnecessarily disruptive of the on-going educational process.
- 6. Dining area (for the staff) separate from the main facility.
- 7. Counselor offices with appropriate conference area to assure privacy.
- 8. Department Head Offices with:
 - appropriate conference area to assure privacy
 - staff message center
- 9. Library and/or Instructional Materials Center with qualified librarians and assistants.

C.3 Implementation of Educational Programs

- 1. An advisory committee made up of teachers, the Curriculum Coordinator, and a principal shall be established by the BSC and the BEU for the purpose of evaluating textbooks and materials for the various disciplines at each grade level. Workbooks and textbooks shall be supplied as individual differences dictate. The BSC shall provide sufficient materials for each child in each discipline. Appropriate pupil workbooks shall be considered as supplementary teaching aids and shall be purchased as expendable supply items each school year, at the discretion of the Curriculum Coordinator and teacher involved. Recognition shall be made not only of the differing needs of individual children, but also for individual schools.
- 2. A flowchart in each discipline for grade K through 8 should be provided by the respective curriculum coordinator for the guidance of teachers.

APPENDIX D: Individual Professional Development Plan for Brookline Educators

Name: Last	First	Middle	Renewal Year
Home Address	City	State	Zip Code
Primary Area		Certificate Nur	mber
District	School	Grade Level(s) Subject(s)
Professional Developm	ent Points Required for Renewal o	f Primary Area 1	20 or 150
Total number of	of PDPs required in content		
My professional growth	n goals (please number):		

Record of Approved Professional Development Activities for Primary Area

Professional Development Activity	Professional Growth Goal (Goal Number)	Content PDPs	Other PDPs	Date Completed

				<u> </u>	
Use additional copies of this form if necessary.					
Educator's Name	me Certificate Number				
BSC/BEU Unit A Contract - 74					

Initial Review and App	oroval	Date
	tional needs of the s	cator's Individual Professional Development Plan is chool and/or district and is designed to enhance the
Supervisor's Name (print)	Title	Signature
First Two-Year Review	W	Date
The signature below indicates treviewed.	that this educator's I	ndividual Professional Development Plan was
Please check one.		
The Plan remains cons		ational needs of the school and/or district.
Supervisor's Name (print)	Title	Signature
Second Two-Year Ro	eview	Date
The signature below indicates treviewed.	that this educator's I	ndividual Professional Development Plan was
Please check one.		
The Plan remains cons	sistent with the educ	ational needs of the school and/or district.
The Plan was reviewed	d and amended.	
Supervisor's Name (print)	Title	Signature
Final Endorsement		Date
•	•	eviewed this educator's Record of Professional are consistent with the approved professional
Supervisor's Name (print)	Title	Signature

APPENDIX E: Alternative Evaluation Process: Professional Growth Project

ALTERNATIVE EVALUATION PROCESS:

PROFESSIONAL GROWTH PROJECT

Submitted to: The Public Schools of Brookline School Committee and

Brookline Educators' Union

Submitted by: The Joint Labor Management Committee

June 1, 2007

TABLE OF CONTENTS

INTRODUCTION	3
ALTERNATIVE EVALUATION PROCESS:	
PROFESSIONAL GROWTH PROJECT	7
APPENDIX A: SAMPLE STRATEGIES/TECHNIQUES	10
APPENDIX B: PROFESSIONAL GROWTH PROJECT PROCESS FORMS	
1. Conference Form #1	12
2. Progress Check Points #2	15
3. Summative Conference – End of PGP #3	17
4. Overall Evaluation #4	19
APPENDIX C: EVALUATORS	20

INTRODUCTION

As part of the negotiations of the last contract the School Committee asked to create a committee to review our current Unit A supervision/evaluation process and make recommendations to the Brookline Educators Union (BEU) and School Committee (SC) for revision. As stated in the contract:

6.8. D. Labor Management Committee. Effective September 1, 2006, a Labor Management Committee shall be established to review and make recommendations for revision to the Evaluation and Supervision process and the evaluation instrument and forms. The Labor Management Committee shall consist of three members appointed by the BEU and three members appointed by the Superintendent. Such Labor Management Committee shall complete its work and report its recommendations to the School Committee and to the BEU on or before June 1, 2007. The parties agree to engage in negotiations regarding changes to the Supervision and Evaluation process and/or to the evaluation instrument and forms during the term of this Agreement. The Labor Management Committee shall consider moving the April 15th date in Article 6.2.I to a later date.¹

The Joint Labor Management Committee (JLMC) was formed in February 2007, and included:

<u>Unit A members (chosen by the BEU President):</u>

```
David Weinstein (grade 1 teacher – Pierce School);
```

Deb Allen (grade 7 Science teacher – Devotion); and,

Julia Speyer (chemistry teacher – BHS)

Administrative members (chosen by the Superintendent):

Jennifer Fischer-Mueller (Deputy Superintendent for Teaching and Learning);

Gerardo Martinez (Principal – Devotion School); and,

Bob Weintraub² (Headmaster – BHS)

<u>Unit B members (chosen together by the Superintendent and BEU President):</u>

Angela Allen (K - 8 Math Curriculum Coordinator);

Sally Dean³ (Vice Principal – Heath School); and,

Gary Shiffman (BHS Social Studies Curriculum Coordinator).

¹ The April 15 date was discussed. It was agreed to keep the current deadline.

² Bob Weintraub attended the April 4 meeting and provided resources for the work.

³ Sally Dean was unable to participate on the JLMC due to conflicts in her schedule.

With individual work in between, the JLMC met for discussion and review on the following days:

April 4, 2007 3:00PM – 4:30PM April 25, 2007 3:15PM – 4:30PM May 3, 2007 8:30AM – 4:00PM May 31, 2007 3:15PM – 4:30PM

Creating the PGP:

Charged with the task of reviewing our current supervision/evaluation process and making recommendations for improvement, the JLMC started with a discussion on the purpose and value of supervision/evaluation. The committee agreed that, at its core, supervision/evaluation is about improving student learning by improving teaching. With that in mind, we discussed the state of our current process.

Descriptions of the current state of supervision/evaluation in the Public Schools of Brookline differed based on each individual's role-perspective and his/her experience in Brookline. There were differences in opinion among the elementary school representatives and between the elementary school and high school representatives. One major point of agreement was that given the current model, evaluators (administrators) are responsible for too many evaluations which may compromise the quality of the evaluations. Therefore, the committee sought to create a process that lowered the number of evaluations administrators were required to complete in any given year, and make the process more "user-friendly" and doable for the administrators, therefore, more effective and meaningful. An additional option is to increase the number of evaluators.

With an understanding of what we wanted the final product to give us, we discussed criteria by which the process created would be assessed⁴. These included, for example:

The process:

1

- 1. Should make the experience of teaching more satisfying and more enjoyable.
- 2. Should provide for choices, alternatives and decision-making by teachers about what they do.
- 3. Should have activities available to study the knowledge base about teaching.
- 4. Should be simple and easy to manage, free from cumbersome paperwork and procedures.
- 5. Should be thought of as a permanent continuous commitment of the district to the development of its people.
- 6. Should itself be evaluated periodically and be open to revision.
- 7. Should produce a written record of individual activities and growth.

⁴ Saphier, J., <u>How to Make Supervision and Evaluation Really Work, 1993</u>, p. 66 – 67.

- 8. Should encourage experimentation in one's teaching.
- 9. Should have adequate resources to make it work.
- 10. Should clearly separate administrative evaluation for job decisions (Comprehensive Evaluation) from the alternative plan.
- 11. Should build in ways for teachers to get support for their growth, their improvements and their achievements.

The JLMC developed this process by reviewing literature on supervision/evaluation, consulting an expert in the field (Jon Saphier), and reviewing processes and documents from other school districts. This process, named the *Professional Growth Project (PGP)*, is recommended for consideration as an option to our current evaluation process for educators with Professional Teacher Status (PTS). This recommendation, for Unit A members of the committee, is contingent upon successful resolution of the issue of time and compensation.

Grounding beliefs and assumptions:

The effectiveness of this alternative to the Comprehensive Evaluation process is based on some beliefs and assumptions.

- In the Public Schools of Brookline, we strive to create and nurture a professional culture which fosters growth through reflection, collaboration, and collegiality.
- Trust is a critical factor Evaluators must trust the intent of the educators and if not, should require a Comprehensive Evaluation. Likewise, educators must trust the evaluators that they are of good will.
- Educators strive to improve their practice.
- Students achieve better in classrooms where the educators adapt their practice to fit students' needs, while still holding high expectations for student learning (McLaughlin and Talbert, 1997).
- The ultimate purpose of professional development is the improvement of student learning.
- Professional development can best be achieved through on-going, consistent support and reflection with colleagues.

Points of Disagreement:

The JLMC created this process collaboratively; we agreed on all aspects of the process as described in the document; however, there are two (2) points of disagreement.

While there is agreement that this process requires educators' time, the committee could not come to agreement on how time would be provided or from where it should come. The Unit A members of the JLMC strongly believe that reallocation of significant current meeting or professional development time and or compensation for additional time will be necessary to make this four year plan work successfully. In the absence of such time and or compensation, the current two year evaluation system should be retained.

The administrators believe that these opportunities and types of activities already exist (faculty meetings, professional development groups) and could happen within these times, not instead of. The administrators felt strongly that this committee should not take the power of deciding how to use faculty meetings away from Principals, the Headmaster, and Curriculum/Program Coordinators.

Also, the JLMC did not agree on the evaluation assignments of K-8 Curriculum Coordinators to pre-professional educators, including grades 7/8 educators and specialists⁵. The Unit A members of the committee believes that *all pre-professional* educators should be evaluated by the building level administrators. The administrators believe that building level administrators should evaluate pre-professional teachers if reasonable within their full load for that year.

With these two (2) significant areas of disagreement, the JLMC is pleased to present an alternative evaluation process for the consideration of the SC and BEU as an addition to the current evaluation process of PTS educators⁶.

_

⁵ BHS Coordinators evaluate all educators within their department, Pre-professional and Professional Teacher Status.

⁶ The JLMC did not discuss the process for educators in the pre-professional phase or the evaluation of Unit B staff.

ALTERNATIVE EVALUATION PROCESS:

PROFESSIONAL GROWTH PROJECT

For the first three (3) years of employment, educators have Pre-professional Status. During this time, the educator is evaluated using the Comprehensive Evaluation described in the Unit A Collective Bargaining Agreement. Following three successful years of teaching in the Public Schools of Brookline, if the educator's employment is continued, s/he receives Professional Teacher Status (PTS) and enters a four (4) year evaluation cycle.

Professional Teacher Status (PTS) Evaluation Cycle:

After receiving PTS, educators enter a four (4) year evaluation cycle.

Year 1	No formal evaluation process
Year 2	Comprehensive Evaluation OR Professional Growth Project
Year 3	No formal process
Year 4	Comprehensive Evaluation OR Professional Growth Project

The Professional Growth Project (PGP) is intended to offer PTS educators the opportunity to engage in reflection, inquiry, and analysis of their teaching and develop new skills and knowledge about their teaching and student learning. Sharing the goal of improving student learning by improving teaching, the PGP is part of the evaluation process as an alternative to the Comprehensive Evaluation for educators with PTS.

For Year 2 and Year 4, the educator may request either the Comprehensive Evaluation or Professional Growth Project. The educator's supervisor (Principal, Headmaster, or BHS Coordinator) may accept the educator's request for a PGP, or the supervisor may deny the request and require the educator to undergo the Comprehensive Evaluation. This decision is entirely the supervisor's. The supervisor is required to give specific reasons if s/he denies the PGP option to an educator.

If an educator requests participation in a PGP, and is approved by the supervisor, the next formal evaluation, two (2) years later, will be the Comprehensive Evaluation. In other words, an educator is to receive a Comprehensive Evaluation at least once within a four (4) year period.

Within the four year cycle at elementary schools, the educator's Principal or Vice Principal will be the evaluator, at least once.

PGPs are created by the educators with feedback from their evaluators. While each project will be unique to each educator's design, all PGPs will meet the following criteria⁷:

⁷ Montgomery County Public Schools, Professional Growth System Handbook, p. 24.

Criteria for Professional Growth Projects:

- 1. The project is connected to student learning.
- 2. The project is based on the educator's goals and the broader building and system goals.
- 3. The project is of genuine interest to the educator.
- 4. The project results in the educator's learning, the application of that learning and/or questions for further inquiry.
- 5. The project and the process exhibit clarity and substance.

In a well-developed PGP, it is clear what the educator intends to do (clarity) and what significant expected outcomes that support student learning are targeted (substance). The PGP must be meaningful to the educator, address his/her interests, and be aligned with the school's goals or support the goals in a related manner. With the core goal to improve student achievement, flexibility in aligning the PGP to the school and system goals allows the educator to develop a PGP that focuses on student learning in that educator's class(es).

Examples of PGPs:

The need to differentiate instruction for students based on their learning styles needs to be extended to the adult learners within the school system. The PGP offers educators the ability to self-design and therefore, self-differentiate their evaluation process based on their needs. Created with the educator's evaluator and meeting the previously listed criteria, the PGPs may take on a variety of forms⁸. Below are some examples of possible processes that could frame an educator's project (*See Appendix A: Sample Strategies/Techniques for a fuller list.*)

- Action Research
- Participating in an inquiry group/Critical Friends Group/study group
- Mentoring a new teacher (within the formal PSB Mentoring Program)
- Collaborative Inquiry
- National Board Certification
- Graduate Course

• Developing curriculum and/or assessments

- Preparing and presenting a workshop/seminar
- Peer coaching/Peer observation
- Developing a portfolio

⁸ Montgomery County Public Schools, Professional Growth System Handbook, p. 24.

Professional Growth Project (PGP) – The Process

- 1. Conference for defining teacher goals and the process.
- 2. Two check-point conferences
- 3. Summative conference
- 4. Overall Evaluation of PGP

1. Conference (Form #1)

The process commences with a meeting with the educator and his/her evaluator. The purpose of this conference is to gain a shared understanding of the goal(s) of the educator's PGP and the process the educator will follow in order to meet the stated goals. At this meeting the criteria described above are used to assess the educator's PGP design. The evaluator's signature indicates acceptance of the PGP.

2. Progress Check Points (Form #2)

Prior to the Progress Check Point conference with the evaluator, the educator completes Form #2. At the conference, the educator describes how the project is going, what is working, what needs to be worked on further, if there are any necessary changes to the plan and identifies what supports may be needed. The evaluator gives feedback on the process. Both parties sign Form #2. The evaluator's signature indicates that Form #2 has been received. There are two (2) Progress Check Point conferences to be held prior to the completion of the PGP.

3. Summative Conference - End of PGP (Form #3)

When the educator finishes his/her PGP, s/he completes Form #3 describing the outcomes of the project, including: what was learned, how this learning may be applied and/or questions for further inquiry, and what impact this learning had or potentially could have on student learning. The evaluator's signature indicates that Form #3 has been received.

4. Overall Evaluation (Form #4)

If all the Professional Growth Project forms (#1 - #4) have been signed off by both parties, the evaluator will deem this evaluation "satisfactory."

Appendix A: Sample Strategies/Techniques

Collaborative Options:		Independent Options:		
	Peer Reflective Conversations		Audio/Videotape Analysis	
	Committee or Task Force Participation		Delivery of Workshops/Courses	
	Delivery of Workshops/Courses Development of Curriculum, Instruction, and/or Assessment Materials Study Groups Action Research New Curriculum Development Participation in Teacher Exchange Collaborative Inquiry New Teacher Mentoring Program Peer Coaching/Peer Observation Team Planning and Teaching Audio/Videotape Analysis		Development of Curriculum, Instruction, and/or Assessment Materials Action Research Professional Visits (to visit programs) Review of Professional Literature Training School-based workshop Out-of-school workshop Conference(s) Writing of an analytic or reflective journal Action Research	
	Critical Friends Group Inquiry Group Study Group		National Board Certification Graduate Course(s) Participation in Teacher Exchange	

Appendix B: Professional Growth Project Process Forms

- 5. Conference Form #1
- 6. Progress Check Points #2
- 7. Summative Conference End of PGP #3
- 8. Overall Evaluation #4

Public Schools of Brookline PROFESSIONAL GROWTH PLAN (PGP) - FORM #1

Name:			Date:
Position:			
School:			
(Circle one)	Year 2 in Cycle	Year 4 in Cycle	e
1. What is your goal for	the PGP?		
2. How does your goal r	relate to the school's g	oals (School Improve	ement Plan)?

3. Which professional development strategies/techniques do you plan to use in your PGP? (See Appendix A: Sample Strategies/Techniques)	
4. List anticipated/needed resources:	

	·	
5.	What is your plan to complete your PGP?	(Include two check-in dates.)
Е	ducator's Signature:	Date:
Ev	valuator's Signature:	Date:
D .	Charl Brief #1 D	
Pr	ogress Check Point #1 Date:	
1		

Public Schools of Brookline PROFESSIONAL GROWTH PLAN PROGRESS CHECK POINT – Form #2

Name:				Date:
Position:				
School:				
Duration of Plan: from		ТО		
(Circle one)	Year 2 in Cycle		Year 4 in Cycle	
1. Regarding your PGP,	, what's working?			
2. What needs to be	worked on?			

3. Are there any changes to the PGP needed? If yes, what are they?			
4. In additional symmetry and day continue the DCD2 If an describe the	vvene out		
4. Is additional support needed to continue the PGP? If so, describe the s	support.		
Educator's Signature:	Date:		
Evaluator's Signature:	Date:		
New Check Point Review (#2) or Final Conference Date:			

Public Schools of Brookline PROFESSIONAL GROWTH PLAN END OF PGP – Form #3

Name:			Date:
Position:			
School:			
Duration of Plan: from		TO	
(Circle one)	Year 2 in Cycle	Year 4 in Cycle	
1. What have you learned	ed?		
2. How are you applying	g your new learning?	OR What questions do	you have for further inquiry/learning?

3. Wh	What impact has this learning had or potentially could have on student learning?				
Educate	or's Signature:	Date:			
Evaluat	tor's Signature:	Date:			

Public Schools of Brookline PROFESSIONAL GROWTH PLAN OVERALL EVALUATION – Form #4

Name:			Date:	
Position:				
School:				
Duration of Plan: from	T	0		
(Circle one)	Year 2 in Cycle	Year 4 in Cycle		
OVERALL EVALUATI	ON			
(Check one)	() Satisfactory	() Unsatisfactory		
Comments by evaluator of above (additional comme	•	or the OVERALL EVA	LUATION determination checked	
Comments by the educator may be attached.				
Educator's Signature: _			Date:	
Evaluator's Signature: _			Date:	

Appendix C: Evaluators All Unit B Administrators

High School

- Headmaster
- Assistant Headmaster
- Deans
- Curriculum Coordinators
- Program Coordinators and Directors

Elementary Schools

- Principals
- Vice Principals
- Program Coordinators
- Curriculum Coordinators
 - ➤ K-8 CCs would first be assigned to grade 7/8 content specific educators and specialists
 - ➤ K-8 CCs may evaluate K-6 teachers at a teacher's requests and with the Principal's approval. Principals will not suggest to the educators to request the CC evaluation; however, the Principal may announce to the faculty clarifying this option.
 - > Success of this model requires the following:
 - A commitment to respect the work load of the CCs, taking into consideration the following:
 - if his/her program is in Program Review and at what phase;
 - other committees s/he serves on; and,
 - FTE (full or part-time); and,
 - how many programs s/he is coordinating
 - o Coordination between the CCs and Principals
 - o If there is a disagreement with the evaluation between the CC and Principal, the decision rests with the Principal